

# **IHAB ABDULRAHMAN SATAM**

# Innovative Brain-Computer Interface Systems: Robotics Control through EEG Signals

Supervisor: Prof. Dr. habil. Róbert Szabolcsi PhD, Dr. habil.', Full Professor

DOCTORAL SCHOOL ON SAFETY AND SECURITY SCIENCES

# DECLARATION

I am IHAB SATAM a student of *Bánki Donát faculty of the Doctoral School on Safety and Security Sciences, Óbuda University*, I hereby declare that this Ph.D. dissertation entitled "*Innovative Brain Computer Interface Systems: Robotic Control Through EEG signals*" was written by myself, except where it is cited in the references or the appendixes. I also certify that this dissertation is an original report of my work, and it has not been submitted anywhere for other qualifications or professional certifications.

Signature: Ihab Abdulrahman Satam Budapest, ...... 20

#### ACKNOWLEDGEMENTS

I would like to thank the people who supported me during this research, and my special thanks and appreciation to my supervisor **Prof. Dr. habil. Róbert Szabolcsi** for his valuable help in guiding me during this research, without his support and guidance this research would not have been possible.

Also, I would like to thank **the staff of Óbuda University**, especially the Doctoral School on Safety and Security Sciences and my colleagues who worked with me and co-authored some of the published papers on the topic, as well as **my family** for their continuous encouragement and support.

Special thanks to **the Hungarian people** and the **Stipendium Hungaricum Scholarship program**, for their amazing hospitality and cooperation. In addition, a special thanks to my friends: **Michal Ludwinek** and **Julius Máté** for their support during this journey. I have many people to thank for being part of this research and I want to pay my respect and gratitude to everyone who made this process possible.

TABLE OF CONTENTS	
List of Figures	
List of Tables	
List of Abbreviations	
1. INTRODUCTION	1
1.1. Background	1
1.2. Basis of Brain Computer Interface	2
1.3. Research Objectives and Scope	2
1.4. Hypotheses	2
1.5. Overview	3
2. LITERATURE REVIEW	4
2.1. EEG Signal Recording	4
2.1.1. Invasive Method	4
2.1.1.1. Advantages and Disadvantages	4
2.1.2. Semi-Invasive Method	5
2.1.2.1. Advantages and Limitations of Semi-Invasive Method	5
2.1.3. Non-Invasive Method	5
2.1.3.1. Advantages and Limitations of Non-Invasive Method	6
2.1.4. Comparison of EEG Recording methods	6
2.2. BCI Application	7
2.2.1. BCI in Robotics	7
2.2.1.1. BCI in Assistive Robotic Devices	7
2.2.1.2. Brain Controlled Wheeled Robots	7
2.2.2. BCI in Disease Inspection and treatment	8
2.2.2.1. BCI for Diagnostic Applications	8
2.2.2.2. BCI in Neurological Rehabilitation	8
2.2.2.3. BCI for Mental Health Application	8
2.2.3. Challenges, Limitations and Future Work	8
2.3.Conclusion	9
<b>3. EEG CLASSIFICATIONS ALGORITHMS FOR BETTER</b>	11
ACCURACY	1
3.1. Brain Signal Generation	13
3.2. Data Set	16
3.3. Problem Statement	19
3.4.Proposed Work	19
3.5.Signal Analysis	20
3.5.1. Preprocessing	20

3.5.2. Feature Extraction	21			
3.5.2.1. Wavelet Transform	22			
3.5.3. Classification	26			
3.5.3.1. ANFIS (Adaptive Neuro Fuzzy Inference System)	26			
3.5.3.2. Supervised Machine Learning	30			
3.5.3.2.1. Support Vector Machine SVM	31			
3.5.3.2.2. Neural Network	31			
3.6. Conclusion	35			
4. BCI FOR CONTROLLING HIGHER DEGREE OF FREEDOM ROBOTIC ARM	37			
4.1. Brain Computer interface	38			
4.2. Robotic Arm	43			
4.2.1. Forward Kinematic Model	44			
4.2.2. Inverse Kinematic Model	45			
4.3. Experimental Work (Controlling The Robotic Arm)	46			
4.4. Controlling a higher degree of freedom robotic arm with few mental commands.	62			
4.4.1. Node-RED	62			
4.5. Conclusion	67			
5. BCI CONTROL FOR WHEELED ROBOT	68			
5.1. Material	68			
5.1.1. Mobile Robot	69			
5.1.2. Method of Controlling the Mobile Robot	70			
5.2. Results	71			
5.3. Conclusion	74			
6. CONCLUSION	75			
7. NEW SCIENTIFIC RESULTS	76			
8. FUTURE WORK AND OUTLOOK	79			
9. AUTHOR PUBLICATIONS	80			
9.1. Publications Related to The Research	80			
9.2. Other Publications	80			
10.REFERENCES				

Table of Figures			
Figure (1) MR images of human Brain	11		
Figure (2) Neuron	14		
Figure (3) Synapse	14		
Figure (4) Neuron potential	15		
Figure (5) Brain Signals	16		
Figure (6) Data recording	17		
Figure (7) Time for one trial	17		
Figure (8) Region of motor imagery	18		
Figure (9) ERD/ERS	19		
Figure (10) Signal analysis stages	20		
Figure (11) Wavelet Decomposition	23		
Figure (12) Probability, QQ plot and Histogram	26		
Figure (13) ANFIS	27		
Figure (14) - Fuzzy set for subject 1	27		
Figure (15) - Fuzzy output	28		
Figure (16) - sub 1 Output	28		
Figure (17) - Accuracy for Both Studies	29		
Figure (18) Machine learning types	30		
Figure (19) Architecture of Supervised Machine Learning	31		
Figure (20) SVM	33		
Figure (21) Neural network 1	33		
Figure (22) Neural Network 2	34		
Figure (23) Comparison of Results	35		
Figure (24) system architecture.	38		
Figure (25) Electrode distribution of BCI on the scalp	39		
Figure (26) Emotive Insight BCI	40		
Figure (27) Emotive Software	41		
Figure (28) Mental Command	42		
Figure (29) Facial Expression	42		
Figure (30) Robotic/Human Arm Model	43		
Figure (32) Kinematic Model	44		
Figure (33) The System Block Diagram	47		
Figure (34) The System Flowchart			
Figure (35) Servo Signal	50		

Figure (36) Subject Training	51
Figure (37) Mean and Energy for EEG signal	52
Figure (38) Mental Command	53
Figure (39) Robot Arm (Base) Movement	53
Figure (40) Low Beta Waves In A Quiet Environment	54
Figure (41) High Beta Waves In A Noisy Environment.	55
Figure (42) Execution Accuracy (HITI Brain)	56
Figure (43) Facial Expression	57
Figure (44) Command Recognition Accuracy (Facial Expression)	58
Figure (45) Elbow Movement	58
Figure (46) Execution Accuracy For Combined Joints Movement	60
Figure (47) Pick and Place (Cont'd).	61
Figure (48) HITI Software Safety Option	61
Figure (49) Node-RED	62
Figure (50) Robot Arm Control	64
Figure (51) Execution Accuracy Of Forward Transition Between Joints	66
Figure (52) Execution Accuracy of Backward Transition Between Joints	66
Figure (53) Execution Accuracy of Combined Movement	67
Figure (54) Alphabot	68
Figure (55) ICC Of Mobile Robot	70
Figure (56) Node-RED Flow for Mobile Robot Control	71
Figure (57) Controlling Mobile Robot using BCI	72
Figure (58) Mental Commands Robot Controlling	72
Figure (59) Execution Accuracy of Mobile Robot	73

List of Tables	
Table (1) Methods of EEG recording	6
Table (2) Brain signal bands	13
Table (3) Accuracy of ANFIS	29
Table (4) Classification Accuracy	34
Table (5) DH Parameter of Robotic arm	45
Table (6) Execution Accuracy for Robotic Arm using HITI Brain Software	55
Table (7) Command Recognition Accuracy	57
Table (8) Joints and Brain Actions.	59
Table (9) Execution Accuracy for Combined Joint Movement	59
Table (10) Execution Accuracy of Joint Switch Control	65
Table (11) Execution Accuracy of Combined Movement	66
Table (12) Execution Accuracy of Mobile Robot	73



# 1. Introduction

The brain is our most valuable and enigmatic organ. It comprises around 100 billion neurons, each generally possessing 10,000 connections that can reach up to one meter. Consequently, it constitutes a complex network that allows us to see the environment. The brain has approximately an equal number of glial cells as neurons, over 700 kilometers of blood vessels, the extracellular matrix, and is encased in clear cerebrospinal fluid, all of which collaboratively sustain the delicate environment of the neurons in a healthy condition. At the whole-organ level, this is already exceedingly intricate; yet this is but a portion of the narrative. At each moment, several processes occur in the brain, including electrical impulses between neurons and the intricate chemical communication that facilitates homeostasis. Given the inherent micro-scale complexity of the brain, a natural methodology for comprehending its physiology and function is provided by homogenized, continuum-based modeling, which emphasizes the large-scale behavior resulting from the aggregation of small-scale contributions[1].

Notwithstanding the remarkable advancements in neuroscience study in recent decades, our comprehension of how the brain facilitates object and action recognition, learning, planning, and language utilization remains incomplete. An enhanced comprehension of cerebral function and cognition will facilitate the management of neurological and psychiatric disorders, while also enabling the creation of novel systems, gadgets, or robots that interact with people in a natural and cognitively compatible manner.

This PhD thesis aims to explore the applications of brain signals in robotics applications, such as prosthetics or Robotic arm, Mobile Robots, and home system Automation. The research plan focuses on designing a Brain computer interface (BCI) that enhances the efficiency, cost effective and reliable control of actuators. The proposed BCI system contains several steps starting from EEG signal recordings, preprocessing, feature extractions and classification. And, at the end transform these signals into commands to control actuators.

# 1.1.Background

Brain-computer interface (BCI) is a communication method designed for identifying a subject's brain intentions and converting them into machine commands to regulate the functions of electromechanical devices[2]. Electroencephalography (EEG) may be the most prevalent noninvasive imaging technology in brain-computer interfaces (BCI). Due to the non-stationary character of EEG, typically resulting from variations in electrode impedance or positioning, as well as factors such as individuals' concentration, tiredness, eye movements, or muscle activity, EEG data demonstrate significant intra- and inter-subject variability. Therefore, a recorded EEG pattern from a subject may not be reproducible from the same person later or from several individuals undertaking the exact same task.

1.2.Basis of Brain Computer interface.

The reason for Brain-Computer Interface (BCI) technology is based on the necessity to establish direct communication channels between the human brain and external technologies[3]. Brain-Computer Interfaces

# DOCTORAL SCHOOL ON SAFETY AND SECURITY SCIENCES

(BCIs) seek to circumvent conventional neuromuscular routes, enabling users to engage with computers only via cerebral impulses. This approach is especially beneficial for those with motor impairments, including those affected by paralysis, ALS, or other physical disabilities, who cannot utilize traditional input devices such as keyboards, mice, or touchscreens. The scientific reasoning is utilizing the brain's electrical activity, predominantly recorded by Electroencephalography (EEG), to understand and decode cognitive states, intents, or directives. Through the analysis and categorization of brain signals, BCIs facilitate the operation of assistive devices, prosthetic limbs, communication systems, or robotic arms, therefore improving the quality of life and autonomy for those with significant physical disabilities[4]. Furthermore, BCIs serve as an experimental framework for enhancing neuroscience and comprehending cerebral functioning, since they investigate the brain's encoding of movement goals, attention, and decision-making processes. The advancement of BCI technology serves not only to enhance human-machine connection but also to elucidate the intricacies of brain processing[5], promote advances in neural rehabilitation, and progress neurofeedback treatments.

1.3. Research Objectives and scope

This PhD thesis aims to contribute to the field of BCI technology. In this work we investigate the implementation of BCI in robotic applications. The research objectives are to design a BCI system that collects the signals non-invasively from the brain, then apply different algorithms for preprocessing, feature extractions and classifications to convert the modified signals into commands to control actuators such as robotic and prosthetic arms, and wheeled robots.

- 1.4. Hypotheses
- **Hypothesis 1**: Due to their simplicity and fewer health hazards, non-invasive BCIs are better for real-world robotic applications than invasive or semi-invasive technologies, notably in assistive robots for disabled users.
- **Hypothesis 2**: The utilization of wavelet transforms for feature extraction, in conjunction with Adaptive Neuro-Fuzzy Inference System (ANFIS) for classification, enhances the level of accuracy of EEG data analysis. The same can imply if Supervised Machine learning algorithms are used for classification.
- **Hypothesis 3**: Employing a Brain-Computer Interface (BCI) for controlling a 6 Degrees of Freedom (DOF) robotic arm that closely replicates human arm movements facilitates more intuitive and precise robotic control.
- **Hypothesis 4**: An innovative technique to control higher degrees of freedom in a robotic arm with less cognitive commands from the brain can improve usability and efficiency. This method enables the users to control a higher degree of freedom robotic arm using 4 mental commands only.
- **Hypothesis 5**: Using Brain-Computer Interface (BCI) technology for the control of mobile robots will facilitate efficient and effective robotic control.

# 1.5.Overview.

In this research, a BCI system was designed in order to use mental commands from the brain in different Human-Robot applications. This proposal is expected to facilitate the complexity of controlling Robots using brain activities. The Human-Machine application has been used for a long time. However, there is always a new method of implementing control that is developed every time.

We have investigated how the BCI system worked and how it can be developed to be more efficient and have higher accuracy. This thesis is organized into 7 main chapters. Each chapter contributes to the understanding of the applications of BCI in human robot interaction applications. Chapter 2 offers a preliminary study that provides insights into the existing knowledge of BCI system and identify the research gap. Chapter 3 presents the methodology and the conceptual design of the BCI system. Chapter 4 contains the experimental work of each hypothesis. Chapter 5 presents the safety and security aspects of the BCI and why the chosen method of the signal recorded was selected. Chapter 6 identifies the results of each hypothesis. Chapter 7 discusses the results and what new scientific contributions have been made in this work.

#### 2. Literature Review

The application of brain signals in robotics is a swiftly evolving domain that investigates the utilization of brain-computer interfaces (BCIs) to provide direct communication between the human brain and robotic systems [6]. Researchers are exploring advanced techniques to convert mental directions into actionable inputs for robotic systems by using electrical brain activity through sensors such as electroencephalograms (EEGs). This method creates new opportunities in areas including assistive robotics, neuro-prosthetics, and human-robot interaction, presenting possible advancements in improving mobility and autonomy for those with motor disabilities [7]. Applications of brain signals concentrate on deciphering diverse mental states such as intention, concentration, or imagined movement utilizing machine learning algorithms that translate these signals into orders for the operation of robotic arms, mobile robots, or exoskeletons [8]. This integration of neurology and robotics seeks to develop intuitive and adaptable systems that can respond to human intents instantaneously. To comprehend the breadth and possibilities of these breakthroughs, it is crucial to analyze fundamental research that explores the principal methodologies of signal gathering, feature extraction, classification methods, and integration with robotic platforms. These first experiments establish the foundation for creating resilient and adaptive brain-controlled robotic systems.

#### 2.1.EEG Signal recording.

Electroencephalography (EEG) is an essential tool in brain-computer interface (BCI) research and several therapeutic applications [9], offering insights into cerebral activity by capturing electrical impulses produced by the brain. There are three main EEG recording techniques: invasive, semi-invasive, and non-invasive [10], each presenting distinct advantages, drawbacks, and uses.

## 2.1.1. Invasive Method

Invasive EEG, often known as intracranial EEG, is a technique in which electrodes are directly inserted into brain tissue or positioned on the brain's surface, behind the dura mater [11]. This method yields superior recordings, accurately capturing intricate brain activity with exceptional spatial and temporal resolution. Invasive EEG includes Depth electrodes implanted in targeted regions of the brain, specifically inside deep brain structures like the hippocampus. They are often used to observe regions that are less accessible to non-invasive techniques and are essential in the management of disorders like epilepsy [12]. Depth electrodes are particularly advantageous in pre-surgical planning, since accurate localization of brain activity is crucial for focused intervention.

#### 2.1.1.1. Advantages and Limitations

Invasive EEG offers the most precise depiction of cerebral activity. The operation is significantly invasive, presenting hazards such as infection, inflammation, and possible long-term damage to cerebral tissue [13]. Thus, invasive EEG is often confined to clinical environments, where the advantages – such as seizure localization for epilepsy management – exceed the surgical hazards.

# 2.1.2. Semi Invasive Method

Semi-invasive EEG holds an intermediary position between invasive and non-invasive techniques. The electrodes are positioned inside the skull, external to the dura mater, enhancing signal quality compared to non-invasive methods while preventing direct contact with the brain [14]. Illustrations comprise:

- 1- The subdural grid and strip electrodes are positioned between the skull and dura mater, providing mild invasiveness and superior signal quality relative to non-invasive methods. They are often used in pre-surgical assessments, especially for epilepsy patients, to identify brain areas associated with seizure activity.
- 2- Epidural electrodes are situated above the dura mater and under the skull, making them somewhat less invasive than subdural electrodes. They provide sufficient spatial resolution while mitigating some dangers linked to direct brain contact. This strategy is sometimes used in research and is beneficial for prolonged observation in certain clinical scenarios.
- 2.1.2.1.Advantages and Limitations of Semi-Invasive Method

Semi-invasive EEG offers a suitable balance between invasiveness and signal fidelity. The electrodes, while not in direct touch with the brain, are positioned inside the skull, yielding sharper data than non-invasive methods. Semi-invasive techniques need surgical intervention, albeit they have a reduced risk compared to fully invasive procedures. This restricted use is often limited to situations, when accuracy is crucial, but a completely intrusive method is considered too hazardous.

# 2.1.3. Non-Invasive

Non-invasive EEG is the predominant and extensively used technique owing to its safety, simplicity, and cost-effectiveness [15]. This method involves placing electrodes on the scalp to assess electrical activity across the skull. The prevalent configuration consists of a cap equipped with several electrodes (from 16 to 256), enabling multi-channel recording across various brain areas [16]. Non-invasive electroencephalography techniques encompass:

- 1- Scalp EEG: Conventional EEG entails the placement of electrodes at defined sites on the scalp, according to protocols such as the 10-20 system, which standardizes electrode positioning for uniform data acquisition. This approach is favored in clinical and research environments due to its accessibility and low danger.
- 2- Dry and Wet Electrodes: Wet electrodes need conductive gel to enhance signal transmission, guaranteeing intimate contact between the skin and the electrode. Dry electrodes, which eliminate the need for gel, have gained popularity due to their ease and mobility, especially in consumer

applications. Despite generating more noise, new advancements have enhanced the reliability of dry electrodes[17].

2.1.3.1. Advantages and limitations of non-invasive methods

The advantages of non-invasive EEG include its great accessibility and safety, with applications in brain-computer interfaces, neuro-feedback, and consumer-grade devices for monitoring and relaxing. However, it also has limitations. Its principal restriction is diminished signal quality resulting from skull attenuation, which reduces spatial resolution and complicates the accurate localization of brain activity. Moreover, non-invasive EEG is susceptible to aberrations caused by ocular movements, muscular activity, and external electrical interference, which might compromise recordings [17].

2.1.4. Comparison of EEG recording methodsTable (1) below compares each EEG recording method according to Application and other different aspects.

Method	Level of Invasiveness	Electrode location	Quality of the signals	Application Scope	Risks
Invasive	High	Brain Tissue	High	Clinical Epilepsy treatment	Infection, Inflammation
Semi- Invasive	Moderate	Between skull and dura mater	High to moderate	Epilepsy Localization, Clinical Research	Mild surgical risks
Non- Invasive	Non	Scalp Surface	Moderate	BCI neurofeedback, Consumer Devices	Minimal

#### Table (1) Methods of EEG recording

Every EEG recording technique has unique benefits and drawbacks, depending upon the intended use and permissible risk thresholds. Invasive and semi-invasive procedures provide superior precision and are crucial in clinical environments where accuracy is vital, such as in the treatment of epilepsy. In contrast, non-invasive EEG is extensively used in research and consumer applications because of its safety and user-friendliness, however this comes at the expense of diminished signal quality. The advancement of sophisticated signal processing and electrode technology is enhancing the capabilities of non-invasive EEG, rendering it more applicable for intricate uses, such as real-time brain-computer interfaces and neuro-feedback systems.

2.2.BCI applications.

The integration of Brain-Computer Interface (BCI) technology with robotics and medical applications is an expanding focus within neuro-engineering, control systems, and biomedical engineering. This chapter examines the existing research environment, fundamental ideas, and technical advancements in BCI applications within robotics and illness diagnosis and treatment [18]. This review analyzes the main approaches, problems, and findings of current investigations, emphasizing the potential of BCI to transform assistive devices, robotic control, and diagnostic systems in healthcare.

# 2.2.1. BCI in Robotics

# 2.2.1.1. BCI and Assistive Robotic Devices

The use of BCI in assistive robots has increased in response to the requirements of people with impairments. Brain-computer interfaces (BCIs) enable people to use robotic equipment, such as prosthetic limbs and exoskeletons [19], via the interpretation of neurological signals, presenting exciting opportunities for enhanced mobility and autonomy. Research demonstrates that BCI-controlled prosthetic arms, often using EEG-based non-invasive BCIs, may attain functionality that closely resembles natural limb motions. Significant advancements encompass:

- Neuro-prosthetic Arms: These devices convert EEG-derived brain signals into accurate motor functions. Advanced systems use machine learning to enhance categorization precision and control dependability. Exoskeleton is a type of assistive robot that can be controlled by Brain signals. They proved to be a great help in assisting disabled people or in rehabilitation for people suffered from strokes, as the work of the teams in[20], [21], [22], [23], [24] show.
- Six Degrees of Freedom (DOF) Robotic Arms: High-degree-of-freedom arms may be manipulated using a restricted array of cognitive instructions, facilitating streamlined but efficient control [25]. Innovative techniques have arisen to enhance command mapping for high degrees of freedom with less cognitive exertion [26].

# 2.2.1.2. Brain Controlled Wheeled Robots

Brain-controlled wheeled robots represent a distinct category of assistive equipment[27], specifically designed for those with mobility impairments. Utilizing BCIs for these applications necessitates advanced signal processing and real-time control techniques. Recent research illustrates the efficacy of wavelet transform-based feature extraction and adaptive neuro-fuzzy inference system (ANFIS) classification in attaining accurate robot control. Investigations in this domain concentrate on:

- Advanced preprocessing and feature extraction approaches, including wavelet transformations, increase signal quality and improve classification accuracy [28].
- Classification using Machine Learning Models: Techniques such as Support Vector Machines (SVM) and neural networks have been used in conjunction with Adaptive Neuro-Fuzzy Inference Systems (ANFIS) to enhance command accuracy, achieving performance levels of up to 90% accuracy.

- 2.2.2. BCI in disease inspection and treatment
- 2.2.2.1.BCI for Diagnostic Applications

BCI has shown promises in illness detection and diagnosis, especially in neurological conditions where non-invasive EEG offers insights into cerebral activity. BCI-based diagnostic systems have been used in the detection of epileptic seizures, the monitoring of neurodegenerative disorders, and the evaluation of cognitive deficits.

- Seizure Detection Systems: EEG-based Brain-Computer Interfaces may be used for real-time seizure detection [29], facilitating early management for individuals with epilepsy [30]. Machine learning methodologies, including neural networks and deep learning, improve detection precision.
- Assessment of Cognitive Impairment: BCI applications in evaluating cognitive functioning may assist in identifying illnesses such as Alzheimer's disease and dementia [31], [32]. Non-invasive EEG equipment, when integrated with cognitive task assessments, has shown their efficacy as significant diagnostic indicators.

# 2.2.2.2. BCI in neurological Rehabilitation

Besides diagnostics, BCI is progressively used in the therapy and rehabilitation of people recuperating from stroke, spinal cord injuries [33], and other neuromuscular disabilities. Rehabilitation robotics, in conjunction with brain-computer interfaces (BCIs), enable patients to manipulate exoskeletons or functional electrical stimulation (FES) devices using their neural signals, therefore facilitating motor rehabilitation.

- Stroke therapy: BCI-driven robots in stroke therapy enable patients to engage in regulated motor motions [34], therefore promoting neuroplasticity and facilitating motor recovery [35].
- Spinal Cord Injury Therapy: Brain-Computer Interface systems integrated with robotic exoskeletons provide paraplegic patients an assistance equipment for mobility training, improving their capacity to regain fundamental motor abilities [36].

# 2.2.2.3.BCI for Mental Health Applications

Brain-computer interfaces are being investigated as instruments for the monitoring and treatment of mental health conditions. Neuro-feedback-based brain-computer interfaces (BCIs) are used to address anxiety, depression, and other mental health disorders by instructing users to modulate certain brainwave patterns linked to stress and relaxation. Research indicates that this method may provide enduring improvements in mental health.

# 2.2.3. Challenges, Limitations and future Work in BCI Application

Notwithstanding its prospective uses, BCI technology encounters several technological and ethical hurdles [37]. Constraints in signal capture, processing, and interpretation provide considerable challenges, particularly in real-time applications. Factors such as auditory interference, user weariness, and the variability of cerebral signals among people might influence the reliability of BCI

performance [38]. Moreover, ethical problems about data privacy and the autonomy of BCI users persist in influencing the development and implementation of these systems. Progress in machine learning, hardware downsizing, and signal processing has the capacity to substantially improve BCI applications in robotics and medical treatment [39]. Innovative technologies, including deep learning for signal analysis, tiny and wearable brain-computer interface systems, and connection with Internet of Things devices, have the potential to revolutionize the use of BCIs in healthcare and assistive technology.

#### 2.3.Conclusion

The use of Brain-Computer Interface (BCI) technology in robotics and medical treatment represents a significant advancement towards individualized, responsive, and user-focused healthcare and assistive solutions. In robotics, brain-computer interfaces (BCIs) have empowered humans, particularly those with significant physical limitations, to operate robotic arms, exoskeletons, and wheeled robots using neural impulses [40]. This advancement has allowed the creation of assistive devices that provide improved mobility, autonomy, and quality of life, along with the possibility of personalized prostheses and highly articulated robotic limbs that closely mimic human motor capabilities. Researchers have enhanced the accuracy and dependability of brain-computer interfaces (BCIs) using sophisticated signal processing methods and machine learning algorithms, hence increasing their accessibility and efficacy for practical applications. In the field of illness diagnosis and treatment, BCIs have shown significant potential in diagnostics, rehabilitation, and mental health therapy. By enabling real-time observation of brain activity, BCIs facilitate the early identification of disorders such as epilepsy and cognitive decline, hence assisting in prompt treatments. For patients in rehabilitation for stroke, spinal cord injuries, or other neuromuscular disorders, BCI-driven robotic devices facilitate neuronal regeneration via regulated motor exercises and functional electrical stimulation. Moreover, BCI neurofeedback systems are shown efficacy in mental health interventions, assisting people in managing illnesses like anxiety and depression by self-regulation of brainwave patterns.

Despite these encouraging developments, BCI technology continues to encounter several hurdles, including signal unpredictability, processing constraints, and ethical issues pertaining to privacy and autonomy. Non-invasive brain-computer interfaces (BCIs) provide a practical, safe, and efficient method for capturing electroencephalogram (EEG) data, making them an optimal selection for both research and application contexts. In contrast to invasive and semi-invasive techniques that need surgical implantation of electrodes, non-invasive BCIs use sensors positioned on the scalp, therefore mitigating the dangers associated with surgery, including infection and tissue damage. This accessibility facilitates wider and more adaptable applications, making them particularly appropriate for the development of assistive technology, neuro-rehabilitation tools, and diverse human-computer interfaces have evolved significantly, with improvements in signal processing and machine learning facilitating precise capture and interpretation of cerebral activity [41]. Furthermore, non-invasive techniques provide prolonged and repeated use without health

hazards, which is crucial for applications in routine contexts such as prosthesis operation, wheelchair maneuvering, and robotic assistance devices. This method integrates safety, user-friendliness, and technical compatibility, making it the favored technique for capturing EEG data in several BCI applications. The literature reviews in the above the sections support **Hypothesis 1**: Due to their simplicity and fewer health hazards, non-invasive BCIs are better for real-world robotic applications than invasive or semi-invasive technologies, notably in assistive robots for disabled users.

# 3. EEG classification algorithms for better accuracy

There are several different components that make up the human brain. These structures include the major cerebrum, the cerebellum, which is smaller, and the brain stem. From magnetic resonance (MR) imaging of the brain, these structures may be identified with relative ease (Figure 1). The cerebrum is made up of two hemispheres: the left and the right. These hemispheres are connected to one another by bundles of nerve fibers known as the corpus callosum connections.

Glial cells and neurons, often known as nerve cells, are the two primary categories of cells that make up the brain. In most cases, a neuron is made up of a cell body, also known as the soma, a long nerve fiber, also known as the axon, and further branching extensions, known as dendrites. White matter and gray matter are the two basic forms of brain tissue matter that manifest themselves as a consequence of the spatial distribution of brain cells. Gray matter is characterized by the presence of neuronal cell bodies and glial cells, whereas white matter is mostly made up of bundles of axons that have been myelinated.

Figure 1 (B) illustrates the distribution of white matter and gray matter in the cerebrum. It shows the white matter as well as the gray matter that is found in the cortical and subcortical regions of the nervous system. The sub-cortical gray matter is comprised of a number of significant structures or areas that are situated deep inside the brain. These include the thalamus, basal ganglia, and the hippocampus [1].







В

Figure (1) MR images of human Brain

The brain is enclosed and shielded by three layers underneath the skull. Arachnoid mater, Dura mater, and pia mater. Based on how it functions, the human brain may be divided into three primary sections. The limbic system oversees emotions, the stem, or cerebellum, is in charge of essential bodily functions, and the cortex is in charge of processing and evaluating critical thought. The frontal, parietal, occipital, and temporal lobes are the four areas that make up the cortex. Major processes including critical thinking, memory storage, visual data processing, sensation, and movement are all controlled by these lobes. The neurophysiology theory states that synapses allow nerve cells to interact with one another. In terms of its functioning, the nervous system is divided into two parts: the peripheral nervous system (PNS) and the central nervous system (CNS). The PNS is made up of sensory neurons that link the body's many sensory receptors to the central nervous system (CNS), whereas the CNS is made up of the brain and spinal cords [1].

In addition, there are two types of cells in the nervous system: neurons, which may send electrical signals, and supportive glial cells, which carry out essential tasks. One particular kind of cell that differs from other types of cells is a neuron. Axons, cell body, and dendrites make up its three components. Cytoskeletal protein and ribosomes, which help in information processing and reception, are abundant in the dendrites. The postsynaptic specialization and the presynaptic terminal make up the synapse. The synaptic cleft, an extracellular gap between these two sections, is where neurotransmitter-mediated communication between neurons takes place. Following its passage through the cell body, the axon is in charge of interpreting the combined signals [42].

In the spinal cord, which is in charge of sending impulses to distant regions parts of the body, the axon length varies from a few microns to up to one meter [43]. As a result, in summary, the body processes signals, the axon transfers signals, and the dendrites receive information. The glial cells' primary job is to keep the neurons connected [44]. Nonetheless, they carry out crucial tasks such maintaining the brain's ionic equilibrium, promoting the healing process following a damage, and regulating the neurotransmitters in the synaptic clef. There are four different kinds of glial cells in the central nervous system: astrocytes, oligodendrocytes, microglia, and ependymal cells. Every one of these kinds serves a crucial purpose [45]. The chemical environment must be balanced by astrocytes in order to improve neuronal signaling. To facilitate quicker signal transport, oligodendrocytes in the central nervous system perform the same role as axons. Microglia helps clear damaged sites of cellular debris. The ependymal cells create cerebrospinal fluid. The action potential (AP) is the name given to the electrochemical signaling that occurs inside neurons [46].

Ion exchange across the neuron membrane is the cause of AP, which is a transient alteration in the membrane potential along the axon. Beginning in the cell body, the AP moves only in one direction. The definition of EEG signal is that it is the measurement of the flowing current during synaptic excitations of the dendrites of neurons in the cerebral cortex. The electrical dipoles between the body of the neuron and the dendrites create electrical potentials. The pumping of positive ions of Sodium  $Na^+$ , Potassium  $K^+$ , calcium  $Ca^+$  and the negative of chlorine  $Cl^-$  causes the current in the brain, and this current generate a magnetic field over the scalp that is measurable by EEG system. There are five types of brain signals regarding the frequency ranges

[47]. These types are Gamma (>35Hz), Beta (12-35 Hz), Alpha (8-12Hz), Theta (4-8Hz) and Delta (0.5-4Hz). Each type of these signals is related to a specific Brain state as shown in Table 2.

No.	Brain Waves	Frequency Hz	Brain Location	Brain State
1	Delta	0.5-4	Frontal Lobe	Deep Sleep
2	Thata	18	Various depends	Light sleep,
2	Theta	4-0	on the state	Meditation
3	Alpha	8-12	Occipital lobe	Relaxation
4	Beta	13-35	Distributed symmetrically	Active thinking
5	Gamma	>35	Somatosensory cortex	High level cognitive function

# Table (2) Brain signal bands

From Table 2, like sleeping or dreaming each band of EEG is generated in a specific state of the mind. Such as sleeping or relaxing etc. as it explained below:

- 1. Delta Waves: These signals are generated in deep sleep, and they are usually very slow waves.
- 2. Theta Waves: These signals are dreamy state, they are generated in sleep and Meditation.
- 3. Alpha Waves: These signals are generated in the rest state of the mind. The waves are noticeable when the eyes are closed.
- 4. Beta Waves: These are generated when the person is active and conscious. The waves can be detected during tasks that require significant attention.
- 5. Gamma Waves: The waves are used for advanced cognitive processing.

The brain signals collected using the Emotive insight have noises and errors [48]. Therefore, the signals need to be preprocessed in order to prepare them for feature extraction that helps with the classification. From Table 2 the Delta and Theta bands are more likely dominant during the unconscious state of the mind, while the Gamma band is dominant in the hyperactive state. For this reason, this work deals with Alpha and Beta for robotic applications.

3.1. Brain Signal Generation

Neurons are building blocks of the neurons system [49]. It consists of the nucleus, axon and axon terminal (Figure 2). The human brain is made up of approximately 60 billion neurons. In a resting state the inside of the neuron is relatively more negative on the outside and this concept is important to realize.



### Figure (2) Neuron

The neurons communicate with each other through an electrochemical signal. Figure (3) shows the synapse. Synapse is the place where the neurons connect and communicate with each other. From Figure (3) the part before the synapse is called the presynaptic terminal or presynaptic neuron and the part after the synapse called the postsynaptic neuron [50]. The presynaptic neuron sends releases the neurotransmitters and they can create potentials on the postsynaptic neurons. This potential is excitatory signaling that leads to an increase in positive charges inside the postsynaptic neurons. It is called excitatory post synaptic potential EPSP. If the presynaptic neurons cause the postsynaptic neurons to become more negative from the inside. It is called Inhibitory Postsynaptic Potential IPSP. Beneath the surface of the scalp, lies the pyramidal neurons. These neurons arranged



Figure (3) Synapse

perpendicular to cortical surface with the apical dendrites closer to the cortical surface. When there is an excitatory postsynaptic potential, there will be an increased positivity inside the cell. The intracellular

positivity leads to a relative increase of the negativity on the outside of the cell which is called the extracellular negativity [51]. The extracellular negativity on one end of the neuron leads to extracellular positivity on the other end of the neuron (the one that is close to the surface). In simple words the negative charge on one end leads to positive charge on the other end. The distant space between the end of the neuron the have the difference in charges between the positive and negative end is called the Dipole. The current flows from the positive end to the negative end (figure (4)).



Figure (4) Neuron Potential

If there is an IPSP inside the neurons (negative charge), this will cause a positive charge on the outside space of the cell. Positive charge on one end leads to negative charge on the other end, and the current will flow from positive side to the negative side. The continuous excitation and inhabitation of these neurons leads to change in the potential at the end of the neurons. This difference in potential for the neurons that are close to the surface of the scalp can be detected using the electrodes [52]as shown in figure (5).



Figure (5) Brain Signals

Electroencephalography (EEG) classification has become a fundamental technique in brain-computer interfaces (BCIs), revealing substantial promise across several domains, including neuroscience, medicine, and assistive robots. EEG signals, produced by cerebral neuronal activity, are intricate, high-dimensional, and susceptible to noise due to their vulnerability to artifacts from muscle movement, external disturbances, and intrinsic non-stationary properties [53]. This unpredictability is a significant difficulty for precise signal classification, as nuanced patterns in the EEG data must be consistently recovered to distinguish mental states, motor images, or cognitive directives. In light of these problems, the development of effective EEG classification algorithms that attain high accuracy has emerged as a primary priority in BCI research, since even little advances in classification accuracy can result in significant improvements in system efficiency and user experience. Recent improvements in signal processing and machine learning have facilitated the development of novel EEG classification methods that enhance accuracy [54]. A principal method is preprocessing techniques, including band-pass filtering and artifact elimination, to enhance signal quality prior to classification. Feature extraction techniques such as Wavelet Transform, which encapsulate both temporal and spectral attributes of EEG data, have demonstrated potential in improving feature clarity and discriminability. Machine learning approaches, such as Adaptive Neuro-Fuzzy Inference Systems (ANFIS), neural networks (NN), and support vector machines (SVM), are utilized for effective categorization, each offering distinct advantages for the task. ANFIS utilizes fuzzy logic concepts to manage uncertainty and unpredictability in EEG data, whereas neural networks, particularly deep learning models, are proficient in recognizing intricate, non-linear patterns. Support Vector Machines (SVM), recognized for their effectiveness in high-dimensional environments, are a favored option for EEG classification, frequently achieving high accuracy with little overfitting. These strategies, utilized either independently or in conjunction, have shown remarkable enhancements in EEG classification accuracy, with current research indicating success rates over 90%, a considerable advancement from earlier methods that frequently failed to surpass 65%. These developments have significant significance, especially in BCI applications designed to aid people with impairments via neuro-prosthetics or in the operation of assistive robotic equipment, where accurate and dependable signal interpretation is essential.

#### 3.2. DataSet

In order to prove the second hypothesis, I used a dataset from a previous study. The data was made by [55] and the ethics board at the Chinese Academy of Science's Institute of Automation approved each trial.

Twenty-five healthy, right-handed people (19 men and 6 women) participated in the poll. The participants lacked knowledge of Mi-Based BCI. To completely comprehend the scenario, it is essential to discuss the technique used to record the signals. The individuals looked at the screen from a distance of one meter while sitting comfortably in a chair with their hands resting naturally on their knees. (Figure 6a). Each trial lasted eight seconds and started with a white circle in the center of the screen for two seconds, as seen in Figure 7. A red circle then flashed for a single second as a signal to assist people focus on the approaching goal. Before the desired reaction was needed, the "Hand" or "Elbow" cue was shown for four seconds. The participants were instructed to visualize carrying out the necessary action with their entire body, not just their eyes, during this time. The participants were told to think about anything they wished while relaxing their limbs.The subjects' right forearms and hands were used to record their EMGs. (Fig 6. 2b) to ensure they weren't operating independently (the EMG signals were eliminated during the EEG preprocessing). The 8s experiment and the fantasy were ultimately stopped by a "break" of 1s. During the interval, the patients were instructed to relax and reduce their muscle and eye movements.



Figure (7) Time for one Trial

Event-related synchronization and desynchronization (ERS/ERD) are associated with alpha (8-13 Hz) and beta (14-30 Hz) motor-related activities. When the action is executed or envisioned, ERD manifests as a decrease in a certain frequency component associated with an increase in brain activity. An increase in a

particular frequency component result in increased frequency sensitivity, or ERS. Occasionally, it can be seen even in the absence of tangible action or intention, as it is associated with the suppression of brain activity. Recent study has identified a robust correlation between a particular brain area associated with sensory perception and the capacity to visualize the movement of certain body parts (as seen in Figure 8). The central dark blue region of the brain, as seen in the figure, governs limb motion. The pale cyan area is crucial for guiding hand movement. The region above the ears is responsible for the movement of the cheekbones and the lips. Motor imagery may induce event-related desynchronization (ERD) in the dominant hemisphere and event-related synchronization (ERS) in the non-dominant hemisphere. Figure 9 illustrates the intensity of brain frequency bands, shown in red.



Figure (8) Region of motor imagery



Figure (9) ERD/ERS

The conventional 10/20 System was employed to acquire EEG data at a sampling rate of 1000 Hz with a Neuroscan SynAmps2 amplifier and a 64-channel electrode cap. The left mastoid served as the reference point for the electroencephalogram (EEG) recordings. Electrode impedances were maintained below 10 k $\Omega$  during the testing.

3.3. Problem Statement

In the field of brain-computer interfaces (BCIs), precise classification of electroencephalogram (EEG) data is essential for efficient command of extraneous devices like mobile robots or robotic arms. The complicated and chaotic nature of EEG data makes it difficult for conventional approaches to achieve high classification accuracy. The practical uses of EEG-based control systems have been severely hampered by previous studies' low classification accuracies. The dataset has a low score of classification accuracy. The goal of the work is to use state-of-the-art machine learning and signal processing methods to improve the EEG signals' classification accuracy. In addition to proving the usefulness of new approaches, this enhancement would make EEG-based control systems more practical and dependable for use in autonomous robotic systems and prosthetics.

3.4. Proposed Work

The data collected was processed using the EEGLAB toolbox (v14.1.1\_b) within MATLAB (R2015a). During the preliminary phases of processing, we utilized a common average reference (CAR). A 40-hertz low-pass filter and a 0.1-hertz high-pass filter were implemented. The input was down sampled to 200Hz to minimize processing expenses. Automatic artifact removal (AAR) was employed to

eliminate anomalies associated with ocular and muscular activity in the EEG. The dataset has already undergone preprocessing in [55] and was prepared for feature extraction. For feature extraction, the wavelet transform was used to accomplish it.

#### 3.5. Signal analysis

The signals collected from the brain are subjected to noise and artifacts due to muscle movements or blinking. For that reason, they need to preprocess to get rid of those deformities. The process starts with a signal preparation phase. This stage is essential since it involves removing a lot of noise and abnormalities from the data. It is necessary to extract the best features from the data and use feature selection to reduce the Dimension. The actual classification process is the final step. The steps in this investigation are shown in Figure 10.



Figure (10) Signal analysis stages

#### 3.5.1. Preprocessing

Preprocessing is an essential first phase in the analysis of electroencephalography (EEG) signals, aimed at augmenting signal quality and enhancing the precision of subsequent tasks such as feature extraction and classification [56]. EEG signals, which record brain activity via electrical impulses on the scalp, are inherently intricate and susceptible to interference from several sources. Sources of noise, including ocular movements, muscular activity, and power line interference, can substantially

impair the quality of recorded EEG data, complicating the analysis and interpretation of the underlying brain signals. To precisely interpret brain activity, efficient preprocessing techniques must be employed to eliminate or reduce artifacts, guaranteeing that the signals analyzed are both pristine and reflective of authentic brain activity [57]. EEG preprocessing often encompasses many phases, each targeting distinct facets of signal quality. Standard procedures encompass filtering, wherein band-pass filters delineate pertinent frequency ranges, and artifact removal, utilizing techniques such as Independent Component Analysis (ICA) to segregate and eradicate extraneous signal sources. Baseline correction, signal normalization, and methods such as notch filtering enhance the data by diminishing the impact of extraneous noise, hence rendering the signals more appropriate for comprehensive analysis. These strategies together seek to reduce variability induced by environmental or physiological causes, hence improving the dependability of the processed signals. Preprocessing is crucial in brain-computer interfaces (BCI), cognitive neuroscience, and clinical diagnostics, as high-quality EEG data are vital for precise interpretation and decision-making. Recent improvements in preprocessing techniques, such as adaptive filtering and machine learning-based artifact rejection, provide novel solutions for addressing ongoing issues in EEG data quality.

#### 3.5.2. Feature Extraction

Feature extraction in EEG signals denotes the identification and isolation of significant patterns or qualities within recorded brainwave data that are pertinent to a given task or study. EEG signals are intrinsically complex, comprising substantial raw data accompanied by noise and superfluous information [58]. The primary aim of feature extraction is to diminish complexity by pinpointing the most useful elements of EEG data applicable for classification, grouping, or control tasks. In the area of EEG signals, feature extraction approaches seek to emphasize particular attributes of the signals, including:

- 1- Time-domain features: These characteristics are extracted directly from the EEG signal in the time domain, including mean, variance, skewness, kurtosis, peak amplitude, zero-crossing rate, and more metrics[59]. Time-domain features are valuable when the temporal characteristics of signals are important [60].
- 2- Frequency-domain features: These characteristics encapsulate the signal's frequency components using methodologies such as Fast Fourier Transform (FFT) or Power Spectral Density (PSD) analysis [59]. Frequency characteristics may represent power in designated frequency bands (such as delta, theta, alpha, beta, and gamma) or the peak frequencies within those bands.
- 3- Time-frequency features integrate both temporal and spectral information, often derived using techniques such as Short-Time Fourier Transform (STFT) or Wavelet Transform (WT). The Wavelet Transform is particularly favored for EEG feature extraction due to its capability to capture the non-stationary characteristics of EEG data [61].

- 4- Spatial characteristics: These pertain to patterns seen across several EEG channels. Techniques such as Common Spatial Patterns (CSP) are employed to identify spatial filters that optimize the variation across classes [62].
- 5- Nonlinear features: These encompass metrics that reflect the complexity of EEG signals, including fractal dimensions, entropy (such as approximation entropy or sample entropy), Lyapunov exponents, and the Hurst exponent [63].

The extracted features serve as input for classification algorithms (such as neural networks, SVM, ANFIS, etc.) to discern various mental states, cognitive activities, or control commands derived from EEG data. Efficient feature extraction is essential since it dictates the efficacy of future analyses and control applications.

#### 3.5.2.1. Wavelet Transforms

The wavelet transform is a method for analyzing non-stationary time-scale data, applicable to EEG recordings [64]. The ability to examine non-stationary signals and decompose them into discrete frequency components over many timeframes is highly advantageous. Utilizing WT, researchers may condense complex biological signals composed of several time-varying data sets into a digestible array of diagnostic criteria [65]. The continuous and discrete Wavelet transform formula are both given in equations (1) and (2).

 $WT_x(a,\tau) = \frac{1}{\sqrt{a}} \int_{-\infty}^{\infty} x(t)\psi * \frac{(t-\tau)}{a} dt$  -----(1) Continuous Wavelet transform

Where a represents scale displacement,  $\tau$  represents time displacement, and  $\psi$  is a wavelet basis function, including Hear, db.Series, Coiflet and so on.

 $WT_x(j,k) = \int x(t)\psi_{j,k}^*(t)dt$  -----(2) Discrete Wavelet transform

The Discrete Wavelet Transform (DWT) restricts the parameters a and  $\tau$  of the wavelet basis function  $\psi(a,\tau)$  to discrete values, representing the discretization of scale and translation [66]. Figure 11 illustrates the decomposition of the discrete wavelet transform (DWT) of the EEG signal x(n). The convolution, utilizing low-pass or high-pass filter coefficients, is a multiplication procedure involving two functions, which is then executed through its own sampling. To down-sample, one must halve the sample signal (reduction). Wavelet signals manifest in two forms: approximation and detail signals. A signal obtained from the convolution of the original signal with a low-pass filter represents a detail. In Figure 11, each output generates a detailed signal D and an approximation signal A, with the latter functioning as the input for the subsequent phase. The major frequency component of the EEG signal dictates the number of levels in the wavelet decomposition.



Figure (11) Wavelet Decomposition

Formula of WT and filter h, is a low pass, can be formulated in the formulation as follows

$$H(z) H(z-1) + H(-z) H(-z-1) = 1$$
 (3)

In the above formula, H(z) is used to represent the h, z-transform filter and the complement transformation of this high-pass filter is expressed as:

$$G(z) = zH(-z-1).$$
 (4)

According to Section above, DWT is used to evaluate the spectrum components of EEG data. EEG signal analysis relies heavily on WT, specifically the careful selection of a wavelet and the optimal number of breakdown stages. The number of thresholds is calculated based on the primary frequency component of the EEG data. In order to classify signals, the levels are chosen such that the wavelet coefficients maintain a strong connection between the various parts of the signals and the requisite frequencies. The analysis was performed using five distinct degrees of decomposition. Therefore, the EEG data is segmented into D1-D5 details and a final method, A5. Multiple wavelet varieties are typically tested to find the most effective combination for a particular application. As a result of its Daubechies wavelet feature, second order (db2) filtering is more adept at detecting variations in the input signal. Therefore, wavelet coefficients were generated using db2 for this study [67]. For the Daubechies wavelet of the second order (db2), the band frequencies are as follows, with a sampling frequency of 256 Hz: D1 (64-128 Hz); D2 (32-64 Hz); D3 (16-32 Hz); D4 (8-16 Hz); D5 (4-8 Hz); and A5 (2-4 Hz). (0 - 4 Hz). To determine discrete wavelet values, MATLAB is used. Because even the most effective classifier will fail with a badly selected input feature, this is a crucial factor in the design of artificial neural networks based on pattern categorization. Determining the wavelet discontinuous coefficient provides a representation of the signal's energy across time and frequency. For this reason, the discontinuous wavelet coefficient calculated from the EEG signal of each record serves as the feature vector used to characterize the

signal. The size of the recovered feature vector is reduced by using statistics on top of the collection of wavelet coefficients. The temporal frequency distribution of the signals under study is represented by the statistical characteristic listed below:

- Means and standard deviation value.
- Variance.
- Skewness.
- Kurtosis.
- Root Mean Square.

The data needed for the right arm is collected from C3 channel. In order to retrieve the features for the EEG data prior to classification, code is developed in MATLAB for this wavelet.

a) Means and Standard Deviation Value

The definition of the mean is very simple as it is the sum of all the signals divided by the number of the signals [68].

$$\mu = \frac{1}{N} \sum_{i=0}^{N-1} X_i - \dots$$
 (5)

The expression  $|X_i - \mu|$  indicates the difference between the deviation of the sample and their mean. The average deviation can be found by the sum of all the derivatives of the sample signals and dividing by the total number of samples. The standard deviation is similar, but the average is done by power instead of amplitude as shown in equation (6).

$$\sigma = \sqrt{\frac{1}{N-1} \sum_{i=0}^{N-1} (x_i - \mu)^2}$$
(6)

b) Variance

It is the variability measure. In order to determine the variance, the average cubed departure from the mean is used as the denominator. The extent of dispersion in a data collection can be better understood by examining its variance. Variance from the mean increases as data spreads out[69].

$$\sigma^{2} = \frac{1}{N-1} \sum_{i=0}^{N-1} (X(i) - \mu)^{2} - \dots - (7)$$

c) Skewness

Skewness is a statistical measure of the degree to which a signal deviates from its mean value. To compute it, divide the cubed standard deviation by the cubed mean variation [70].

$$\gamma = \frac{1}{(N-1)\sigma^3} \sum_{n=0}^{N-1} (x_n - \mu)^3 - \dots$$
 (8)

d) Kurtosis.

It is the Kurtosis of the signal that determines its Peakedness. More peaks in the waveform correspond to a greater kurtosis number [70].

$$K = \frac{\frac{1}{N-1} \sum_{i=0}^{N-1} (x_i^4)}{(\frac{1}{N-1} \sum_{i=0}^{N-1} (x_i^2))^2} \dots (9)$$

## e) Root Mean Square.

It is a quantitative representation of the signal's intensity. The signal's magnitude is determined using the root-mean-square formula. The strength is represented by the range [71]. The root-mean-square deviation provides a measure of the variability in the system's response to external factors.

$$R.M.S = \sqrt{\frac{1}{N-1}\sum_{i=0}^{N-1}x_i^2} - \dots + (10)$$

Once the data had been cleaned up, the wavelet transform technique was used to pull out the most useful information. Subject one's characteristics are displayed in a, likelihood, quantile-quantile and histogram figure (a, b, and c, respectively, in Figure 12). The QQ diagram demonstrates that the feature data is normally distributed.





Figure (12 A,B,C) Probability, QQ plot and Histogram

# 3.5.3. Classification

The classification of EEG signals entails the categorization or labeling of various EEG data segments according to distinct mental states, cognitive processes, or environmental stimuli. The objective is to convert intricate brainwave patterns into significant classifications for diverse applications such as medical diagnosis, Brain-Computer Interfaces (BCIs), cognitive state assessment, or robotic control [72]. In this thesis, different algorithm used for the classification of EEG signals.

3.5.3.1. ANFIS (Adaptive Neuro Fuzzy Inference System).

The names "adaptive neuro-fuzzy inference system" (ANFIS) and "adaptive network-based fuzzy inference system" denote artificial neural networks based on the Takagi-Sugeno fuzzy inference system [73]. This method emerged in the early 1990s. By integrating components of neural networks with fuzzy logic, it may harness the advantages of both within a cohesive framework. It can learn and approximate nonlinear functions by a reasoning process akin to a series of IF-THEN fuzzy rules. Consequently, ANFIS is regarded as a global predictor. The ANFIS may be utilized more efficiently and effectively with the ideal configurations identified by a genetic algorithm. Potential applications encompass intelligent energy management systems with contextual awareness. The network architecture has two primary components: the foundation and the outcome [74]. The structure has five increasingly deeper levels. The first layer utilizes the input integers to select the appropriate membership functions. It is frequently termed the "fuzzyfication layer". The membership degrees of each function are calculated using the parameter set {a, b, c}. The second level is responsible for generating the rule-based discharge rates. The second layer is tasked with generating the regulated discharge intensities. The second layer is referred to as the "rule layer" due to the regulations it encompasses. Fuzzyfication introduces an additional layer of complexity. Layer 4 seeks to normalize the projected firing strengths by dividing each value by the overall firing strength. The fifth layer receives the normalized data and the parameter set  $\{p, q, r\}$  as input. The output is sent utilizing the defuzzyfied values provided by this component. Figure 13.

Following feature extraction from the EEG data, ANFIS was implemented using MATLAB code for this research. Subsequent to the incorporation of registration functionalities. The results for each subject were obtained utilizing the previously established FIS procedure.



Figure (13) ANFIS

Figure 14 shows the Fuzzy inference system for the first subject and how the rules are based in order to collect the output.



Figure (14) Fuzzy set for subject 1.

Figure 15 depicts the real result of the fuzzy system, which is the application of fuzzy principles to all the samples within each topic. The outcome changes with the characteristics chosen by WT transform, as demonstrated by the findings. Figure 16 displays the discrepancy between the real and ideal outputs. The


ANFIS algorithm produces results that are within a tolerable margin of error in comparison to the intended results.

Figure (15) Fuzzy output



Figure (16) sub 1 Output

No	Accuracy of ANFIS classification algorithm						
1	85.00%	14	91%				
2	95.30%	15	85%				
3	86.06%	16	93%				
4	96.40%	17	90%				
5	99%	18	87%				
6	80%	19	95%				
7	91%	20	80%				
8	88.10%	21	92%				
9	88%	22	95%				
10	80%	23	96%				
11	85%	24	95,00%				
12	90.30%	25	96%				
13	94%						

 Table (3) Accuracy of ANFIS

Figure 17 given below shows the accuracy achieved in this study compared to the results in [55]. Due to its superior performance compared to its predecessors, the algorithm is heavily relied upon for signal classification.



Figure (17) - Accuracy for Both Studies

#### 3.5.4. Supervised Machine Learning

Machine Learning (ML) algorithms are systems that can discern latent patterns in data, forecast outcomes, and enhance performance [75]. Various methods utilized in MI are illustrated in Figure 18. The picture illustrates three categories of algorithms: Supervised, Unsupervised, and Reinforcement Learning.



Figure (18) Machine Learning Types

Supervised Learning, as the term suggests, requires external supervision for the machine to acquire knowledge. Supervised learning models are taught using labeled datasets. Upon completion of training and processing, the model undergoes testing by supplying sample test data to evaluate its predictive accuracy. Supervised learning is categorized into classification and regression. Unsupervised learning is a method in which the computer learns from data without requiring external supervision [76]. The unsupervised model may be derived from an unlabeled dataset that is neither classed nor categorized, requiring the algorithm to operate on this data without supervision. The categories of unsupervised algorithms include Clustering and Association [77]. Reinforcement learning is a paradigm in which the agent engages with the environment through actions and acquires knowledge via feedback [78]. The primary concept of Supervised Machine Learning is the correlation between input and output data. To do this task, the algorithm is provided with training data. The training data comprises input-output pairs. Inputs are multidimensional vectors that encapsulate pertinent information on the signal states, specifically the brain signal states, or activities to be decoded. Raw data is often utilized to generate features, which are subsequently refined through feature engineering to discern the most promising or pertinent ones. Training encompasses the acquisition of the relationship between attributes and intended results. The response variable, or dependent variable, denotes

the output of interest associated with these characteristics, such as brain state or behavior. In the training phase, the model learns to correlate input features with target variables by optimizing its parameters, which is accomplished by minimizing a cost function [79]. The model's performance is represented by the loss or error assessed by the cost function. Numerous techniques exist that can reduce the cost function, with gradient descent being the most prevalent. An appropriate machine learning algorithm model comprises training and testing datasets. The testing set must not be assessed by the algorithm during the training phase. It must precisely represent the model's true environment. The outcomes of the testing sets should be comparable to those of the training sets. Figure 19 illustrates the machine learning architecture.



Figure (19) Architecture of Supervised Machine Learning

There are several types of supervised machine learning. In this thesis, the Support vector machine and neural network were used.

#### 3.5.4.1.Support Vector machine (SVM).

Support Vector Machine (SVM) is a robust and versatile supervised machine learning method mostly employed for classification tasks, while it may also be applied to regression. Its capability to manage both linear and non-linear data significantly enhances the performance of several challenging classification jobs [80]. Support Vector Machines (SVM) often identify the most suitable separating border, whether linear or hyperplane. Support Vector Machine (SVM) identifies the hyperplane that maximizes the margin for a binary classification problem, signifying that it is the furthest distant from the nearest data points of each class. Support vectors, the nearest points, are crucial for defining the hyperplane [81].

#### 3.5.4.2. Neural Network (N.N).

A neural network designed for optimization or enhancement over time is termed a "optimizable neural network." The optimization of a neural network often aims to improve its efficacy in certain tasks, such as

predictive modeling, language processing, or picture recognition [82]. The following are essential principles pertaining to optimizable neural networks:

- 1- The learning algorithm of an optimizable neural network is its most critical component. This method, often a variant of gradient descent, adjusts the network's weights based on input. The aim is to minimize the discrepancy, sometimes termed as the cost or loss, between the network's projected and actual outcomes.
- 2- Backpropagation is a prevalent optimization method utilized in neural networks, especially within deep learning architectures. It is a method for rapidly computing the gradients of the loss function with respect to the network weights. The weights are subsequently adjusted utilizing this information to minimize the loss.
- 3- Hyperparameters: These are the configurations or settings that govern the neural network's overall behavior but remain unchanged during the learning process. Examples include learning rate, batch size, and the number of network layers. Fine-tuning hyperparameters is a crucial step in optimizing neural networks.
- 4- Overfitting occurs when a network becomes overly specialized to the training set, resulting in worse performance on new, untested data. Regularization techniques like dropout, weight decay, and early stopping mitigate overfitting, hence enhancing the network's generalizability and performance in realworld applications.
- 5- Data Preprocessing: The efficacy of the network can be significantly influenced by the method of data preparation and presentation. Normalization, standardization, and augmentation are strategies that enhance the efficacy and efficiency of network training.
- 6- Transfer Learning: Employing a pre-trained model on a substantial dataset and subsequently fine-tuning it for a specific task is a prevalent technique utilized to improve neural networks. This method can yield significant performance improvements, especially when there is a scarcity of data for the specific task.
- 7- Evolutionary Algorithms: Advanced approaches such as evolutionary algorithms are employed to optimize neural networks. These strategies iteratively improve network performance by generating a population of networks, selecting the highest-performing ones, and utilizing them to create a new generation of networks.
- 8- Hardware Optimization: Finally, utilizing GPUs (Graphics Processing Units) or TPUs (Tensor Processing Units) for accelerated processing a vital component in the efficient training of large networks can also improve the performance of neural networks at the hardware level.

The featured extracted in section 3.5.2.1 helps the classification algorithms to train in order to find a pattern that can predict the outcome for each trial. The data was labeled as 1 and 2 which represent if the subject imagined moving his elbow or wrist. Figures 20 and (21-22) show the SVM and NN algorithms respectively for the classifications.

LEARN TEST EXPLAIN														2
New Save + Feature Optimizer	All Quick-To- Train	AB MODELS	All Linear	-	Use Parallel	Train All -	Scatter	Confusion Matrix	Res Ta	ults Leyout ble *	Export Plot to Figure	Generate Function EXPORT	Export Model •	
Models !	Model 1 M	lodel 2						100.010401	1.301.13			610 0151		I
Sort by (Model Number 🔹 🗐 🗍 📰 👘 👘	Summary × 1	Minimum Clas	soffication Erro	r Plot	Valida	tion Cont	usion Matrix	×						
() 1 Tree Accuracy (Validation): 20.0%													-	
Z SVM Accuracy (Validation): 90.0%     ast change: Optimizable SVM: 3/16 features (PCA or	1				M	odel 2				1	Numbe     True P     False f	ir of observation ositive Rate	ations s (TPR) tes (FNR)	
	1 Glass	100.	0%					100.0%			O Positiv False ( What is the	e Predictive Discovery R Confusion	Values (P ates (FDR Lmatrix?	×V)
	2 2	20.0	1%			1		80.0%	20.0%					
		1			2 Predic	ted Cla	is	TPR	FNR					

# Figure (20) SVM



Figure (21) Neural network 1



Figure (22) Neural Network 2

The figure shows the results for only one subject. Table 4 present all subjects results after applying the same signal processing, feature extraction and classifications.

Subject	Αссι	iracy	Subject	Accu	iracy
No.	SVM	NN	No.	SVM	NN
1	90.00%	80.00%	14	85.00%	81.00%
2	85.00%	84.00%	15	73.00%	86.00%
3	85.00%	82.00%	16	78.00%	90.00%
4	80.00%	80.00%	17	88.00%	85.00%
5	84.00%	81.00%	18	84.00%	81.00%
6	82.00%	82.50%	19	86.00%	83.00%
7	75.00%	84.00%	20	91.00%	81.00%
8	80.00%	83.00%	21	96.00%	81.00%

 Table (4) Classification Accuracy

9	79.00%	84.00%	22	98.00%	81.00%
10	76.00%	87.00%	23	85.00%	85.00%
11	85.00%	80.00%	24	96.00%	82.00%
12	80.00%	80.00%	25	82.00%	81.00%
13	77.00%	80.00%			

Figure 23 shows the difference between the SVM, and NN results compared to the results shown in [83]. The improvement made in this study shows a good step up in classifying the signals.



Figure (23) Comparison of Results

#### 3.6. Conclusion

The focus of this chapter is to identify a new method for EEG signal analysis to achieve higher accuracy. In this work, the study from [55]was chosen for the investigation. The following points were noted.

- 1- Daubechies wavelet feature, 2<sup>nd</sup> order (db2) filtering is more adept at detecting variations in the input signal. Therefore, wavelet coefficients were generated using db2 for this study.
- 2- fuzzy set theory is useful for addressing ambiguity and making choices. With the help of fuzzy logic, we were able to incorporate doubt into the classifier's architecture. The categorization of EEG data. When the wavelet coefficients of the EEG signals were used as inputs, the ANFIS algorithms were able to distinguish between two groups of EEG signals. Although it will take a long time to process, this can be overcome by using a computer with a high CPU process, the more characteristics pulled

from the data, the more efficient the program can be. When using the algorithm to control the motion of a mechanical arm, the varying results increase the likelihood that the arm will be moved by varying the pace at which the elbow and the wrist rotate. The ANFIS was assessed based on the classification outcomes and data metrics. The overall precision of the ANFIS model's classification was 90.13%. Indicating the algorithm's dependability and potential for further uses, the suggested ANFIS model can be used to classify EEG data.

3- The identical approach was demonstrated using the Supervised Machine Learning algorithm for the classification accuracy of brain signals. The results illustrate that the algorithms enhanced the categorization accuracy of the previous study, suggesting that the research was progressing appropriately. Attaining this level of accuracy facilitates the formulation of novel strategies for future enhancements in precision. Various methods exist for categorization; however, in our investigation, we utilized only two, as they yielded the most favorable outcomes. The results were derived from the characteristics retrieved. The findings of the two methods of classifications support Hypothesis 2 that indicates " The utilization of wavelet transforms for feature extraction, in conjunction with Adaptive Neuro-Fuzzy Inference System (ANFIS) for classification, enhances the level of accuracy of EEG data analysis. The same can imply if Supervised Machine learning algorithms used for classification".

#### 4. Using BCI for Controlling Higher Degree of Freedom Robotic Arm

The amputation of limbs significantly impacts the individual's everyday activities. Individuals with upper limb loss encounter difficulties in tasks such as tossing a ball, operating a vehicle, and engaging in handshakes. The concept of regulating actuators, including robotic and prosthetic limbs, mobile robots, automobiles, and bicycles [84]. The use of brain signals has advanced significantly during the past few decades. Nevertheless, more investigation remains to be conducted. The procedure has many stages. The initial step is to non-invasively record signals from the subject's brain. Secondly, it is necessary to preprocess these signals by eliminating any artifacts and noise to prepare them for the subsequent classification phase. Subsequently, we may utilize these signals to translate them into commands for actuators, contingent upon the specific applications in which they are employed. The Brain-Computer Interface (BCI) facilitates communication between the brain and computers by capturing electrical activity and connecting external devices. A Brain-Computer Interface (BCI) is a device that detects and collects electroencephalogram (EEG) signals generated by the brain and translates them into orders to execute a specific action for an externally linked device. The non-invasive technology is economically viable and possesses significant promise for many human-robot applications, including wheelchairs and prosthetics [85]. The concept of BCI involves gathering signals from the scalp, which are then transformed into a functional output. The microcontroller is controlled by this active function. The microprocessor assesses the incoming signals to ascertain the degrees of freedom, prompting the robotic arm to respond accordingly. The robot kinematic model pertains to the robot's mobility independent of the forces that generate it. As mentioned in chapter 3, the bands of EEG signals are classified in 5 types. For motor imagery, the Alpha and beta bands were used for robotic application. The system component can be seen in Figure 24. The figure shows the system component and bifunctional feature of each part of the system.



Figure (24) system architecture.

## 4.1. Brain Computer Interface.

BCI is a device that makes direct connections between electrical activity from the brain and an external device [86]. For recording the signals, there are typically three methods. The invasive, semiinvasive and non-invasive method as explained in chapter two. Due to the easement of installations and it does not require any incisions or stiches, the non-invasive method is used in this work. The device used in this work is Emotive Insight 2.0 with 5 channels. The device has 5 EEG sensors to detect brain activities and two reference (CML/DRL) sensors. One of the reference sensors is an active electrode (CML: Common Mode Sense) and the other is Passive electrode (DRL: Driven Right Leg).



Figure (25) Electrode distribution of BCI on the scalp

As shown in Figure 25 the yellow circles represent the location of the reference sensors, and the green circle represents the sensor locations, which are the frontal, temporal and parietal part of the head. Emotive insight provides a non-invasive solution for detecting brain signals. The device (as shown in Figure 26) detects a wide range of brain signals from three different parts of the brain, the Frontal lobe specifically from points (AF3) and (AF4), the Temporal lobe by points (T8) on the right and (T7) on the left and last the Parietal lobe from point (Pz). The device's measurements are based on six keys: focus, Stress, Excitement, Relaxation, interest, and engagement [87]. The Emotive Insight software is an open-source software that can be downloaded from the Emotive website. There are several versions of the software, each one is used for different applications. Emotive BCI is the one that is used in this work. The Emotive BCI is a desktop application. It can be used in both Mac and Windows. The Emotive BCI allows the user to view and train the brain. The data streams for the Emotive BCI are classified in four categories: Mental Commands, Performance Metrics, Facial Commands and Motion sensors.

- Mental Commands: the software allows the user to train the brain based on commands such as Push, Pull, move right, Move left.... etc.
- Facial Expression: the software can trigger events based on facial expression. It records the brain signals based on smile, blink, wink .... etc.



Figure 26 Emotive Insight BCI.

The EEG signals are passing through a few stages of preprocessing. Firstly, the data is processed to remove sharp spikes. And then passed to a high pass filter to remove the DC offset and slow drift. The Emotive insight uses 2-second epochs and applies a Hanning-filter before performing the FFT. The POWER is calculated from the square of the amplitude in each frequency bin and output as  $uV^2/Hz$ .

The emotive software has the option of displaying all the signals detected from the brain. The brain data is recorded from five points (as the device used is emotive insight 5 channels). Figure 27 shows the Emotive software.





The data sampling of Emotive Insight is 128 HZ with 16-bit resolution. The value of these data is affected by the different thoughts of the person. The emotive software also has the ability to train the brain to do some mental commands like: Push, Pull, move left, right etc. as shown in Figure 28.



Figure (28) Mental Command

It also can record the signals of the brain due to facial expressions such as: smile, Blink, Wink left and right etc. (Figure 29). These records are useful to use for further applications in moving mobile robots or controlling multi degree of freedom robotic arms.



Figure (29) Facial Expressions

#### 4.2. Robotic Arm

The 6 DOF robotic arm has a wide range of applications in industry or for personal use as it is the closest to the human arm. The Degree of Freedom (DOF) refers to the number of independent movements the robot arm can make [88]. A 3DOF robot arm can move up, down, right, left, in and out. The more Degree of freedom the arm has the wider range of motions it can perform. The 6 DOF robotic arm can perform three linear motions (forward/back, right/left, up and down) and three rotational movements (tilting, twisting and rotating the arm itself) [89].

This allows for a great precise and complex manipulation, mimicking the flexibility of the human arm. One of the most important steps for controlling robot manipulator is to implement the complete and accurate system's mathematical model. The arm basically is a series of manipulators with revolute joints. The geometric configuration of the arm is made up of base, shoulder, elbow, and wrist. Figure (30) shows the robotic arm in correspondence to human arm model. Each joint except the wrist has a 1 DOF. The wrist has 2 DOF as it can move vertically and rotationally. Each DOF in the arm is actuated by a servomotor. The end-effector is a two-finger gripper. Each joint is connected to a servo motor that allows the joint to perform two movements. 6 servo motors lead to 12 movements for the arm. The combination of brain thoughts enables the arm to perform complex motions such as Pick and Place.



Figure (30) Robotic/Human Arm Model

4.2.1. Forward Kinematic Model.

The kinematic model in this work is achieved using Denavit-Hartenberg (DH) parameter. Figure 31 presents the kinematic model of the robotic arm in L position. The Base, shoulder and elbow are moving the tool point to its desired position, the orientation of the end-effector is implemented by the wrist joints. DH works with twist angle  $\alpha_{i-1}$ , link length  $a_{i-1}$ , link offset  $d_i$  and joint angle  $\theta_i$  {  $\alpha_{i-1}, \alpha_{i-1}, d_i, \theta_i$  }[90].

As shown in Figure 31 a coordinate system is attached to each link of the manipulator. Table 5 lists the DH parameter of the robotic arm. The derivation of the links (joint expressed as i in its previous neighboring joint i - 1 was derived and presented in Equation (1) which represents the overall matrix of the end-effector to the base of the robotic arm.



Figure (32) Kinematic Model

	Table (5) D	)H pa	rame	otic a	rm				
	Symbols		Joints						
	Symbols	1	2	3	4	5	6		
	$\alpha_{i-1}$	0	-90	0	0	-90	0		
	<i>a</i> <sub><i>i</i>-1</sub>	0	0	$L_2$	$L_3$	0	0		
	$d_i$	$L_1$	0	0	0	0	$L_4$		
	$\boldsymbol{\theta}_{i}$	$\boldsymbol{\theta}_1$	<b>θ</b> <sub>2</sub> - 90	$\boldsymbol{\theta}_3$	$oldsymbol{ heta}_4$	$\boldsymbol{\theta}_{5}$	0		
[ C	$C_1 C_5 S_{234} + S_5$	$S_1 S_5$	$-C_1$	S <sub>234</sub> .	$S_5 + C_5$	$C_1 C_{234}$	$C_1$	$C_{234}$	(
${}^{0}_{CT} = \begin{bmatrix} -1 \\ -1 \end{bmatrix}$	$S_1 C_5 C_{234} -$	$C_1S_5$	$S_1$	$C_{234}$	$C_{5} + C_{5}$	$C_1C_5$	$S_1$	C <sub>234</sub>	
61 -	$C_{234}C_{5}$			-0	$S_{234}S_5$		$-S_2$	<sub>234</sub> B	
L	0				0			0	

where

$$A = L_2 S_2 + L_3 S_{23} + L_4 C_{234}$$
  
$$B = L_1 + L_2 C_2 + L_3 C_{23} - L_4 S_{234}$$

From equation (1), the 3\*3 matrix (the first three rows and first three columns) is the rotation matrix. The last column represents the position (*x*, *y*, *z*) of the end-effector with respect to the base.

4.1.1. Inverse kinematic Model.

In practical robotic systems, inverse kinematics have more potential applications. With inverse kinematics, the computation of the joint angle required for achieving the required position and orientation can be calculated [90]. In inverse kinematic, the joint angles  $(\theta_1, \theta_2, \theta_3, \theta_4)$  must be calculated, while  $\theta_5$  is directly given by the desired orientation for object manipulation.

The transformation matrix from the tool to base is given by:

$${}^{Base}_{Tool}T = \begin{bmatrix} n_x & o_x & a_x & p_x \\ n_y & o_y & a_y & p_y \\ n_z & o_z & a_z & p_z \\ 0 & 0 & 0 & 1 \end{bmatrix}.$$
(2)

The first 3\*3 matrix represents the rotation, and the last column represents the translation of the end-effector with respect to the base.

$$\theta_1 = Atan2(p_x, p_y)$$
(3)
  
 $s_{234} = c_1 a_x + s_1 a_y$ 
(4)

$$c_{234} = a_z \tag{5}$$

$$\begin{aligned} \theta_{234} &= Atan2(s_{234}, c_{234}) \quad (6) \\ c_3 &= \frac{(c_1 p_x + s_1 p_y + l_4 s_{234})^2 + (p_z - l_1 + l_4 c_{234}) - {l_2}^2 - {l_3}^2}{2l_2 l_3} (7) \\ s_3 &= \mp \sqrt{1 - c_3^2} \quad (8) \\ \theta_3 &= Atan2(s_3, c_3) \quad (9) \\ c_2 &= \frac{(c_1 p_x + s_1 p_y + l_4 s_{234})(c_3 l_3 + l_2) - (p_z - l_1 + l_4 c_{234})s_3 l_3}{(c_2 l_3 + l_2)^2 + s_3^2 l_2^2} (10) \end{aligned}$$

$$s_{2} = \frac{(c_{1}p_{x}+s_{1}p_{y}+l_{4}s_{234})(s_{3}l_{3})-(p_{z}-l_{1}+l_{4}c_{234})(c_{3}l_{3}+l_{2})}{(c_{3}l_{3}+l_{2})^{2}+s_{3}^{2}l_{3}^{2}}(11)$$
  

$$\theta_{3} = Atan2(s_{2}, c_{2}) \qquad (12)$$

$$\theta_4 = \theta_{234} - \theta_2 - \theta_3 \tag{13}$$

## 4.2. Experimental work (controlling the robotic arm).

The experiment implemented in two conditions, the first one is quiet environment and the second is noisy environment (such as two people were talking next to the subject). The system consists of three different software working together. The Emotive software (for Emotive BCI), the Arduino UNO software and the HITI brain software that combines the two previous software together. The HITI brain is software that can make communication between Arduino board and Emotive EEG headset very easy. It allows the Arduino to receive mental commands from Emotive insight headset. Figure 33 shows the system block diagram.



Figure (33) The System Block Diagram

The target movement in this work is as follows:

- 1- Test all the movement using the mental commands separately.
- 2- Make the full pick and place by combining different mental commands.

As mentioned earlier, the 6 DOF robotic arm is very close to the human arm. The work of the robotic arm is more important than the outside shape of the robotic arm. Therefore, the main aim in this work is the actuators that move the robot links together (aka. The servo motors). The Arduino code defines the motors work and posterization of which motor work is defined in the code according to the mental commands. Figure 34 shows the flowchart of the system working. The actuators' movement is based on the predicted category. The three-software cooperating between each other in order to make a language that the robots understand. The BCI read the signals using the electrodes connected to the cortex, then transfer the signal to the Emotive software for recording and further process for noise reduction and classification. The Arduino UNO has the codes for connects the Arduino code and the Emotive software, executes the code with the database from the emotive software, and transforms it into actual motion to the connected arm. The software working principle is building a GUI to control the actuators according to the code written in Arduino IDE. It's important to mention that the subject needs to be trained before the final test. The emotive software has a training feature through a designed

experiment. The subject sits in a relaxed position in front of the screen. A cube in the center of the screen appears and the subject needs to imagine moving it forward, backward, right, left, up and down. This experiment is important for two reasons. First, is to sharpen the brain to produce the same pattern of signals for each movement. Secondly, store these signals for referencing and compare them with the real time signals in the final test, so the HITI brain can command the Robot arm to move according to the mental command with the help of the Arduino code.



Figure (34) The System Flowchart

The servo motor is controlled by a pulse width modulated signal [91]. The Arduino sends a pulse of 1mS and 2mS long every 20ms to the servo motor. Figure 35 shows the servo signal.



Figure (35) Servo Signal

The servo motor rotates the shafts to the central position during the 1ms pulse. Several pulses rotate the motor enough to move the links to the desired location. The servo library in Arduino generates the necessary PWM signals to send it to the servo. During Motor imagery (imagining movement without actual physical motion), a distinct pattern is shown in the EEG signals in the Alpha and Beta bands. These patterns play a great role in decoding mental commands to control the robotic arm or other devices. In the state of motor imagery task, the Alpha power tends to decrease in the sensorimotor cortex. This is called Event-Related Desynchronization (ERD). The Brain tends to reduce the Alpha-band to prepare for motor actions. On the contrary, during motor imagery, the Beta wave is increasing [92]. This increase is called Event-Relate Synchronization (ERS). The ERS occurs as a response to imagining moving a part of the body [93]. Before the controlling of the robotic arm, the Emotive EEG signals have to be passed through several steps, preprocessing, Feature Extraction and classification.

## A- Preprocessing

As mentioned before, the EEG signals contain noises and artifacts that are caused by Eye blinking, moving the face muscles. Using band pass filter is essential to remove all other bands and focus only on Alpha (8-12 Hz) and Beta (13-30 Hz). As for the artifacts, the Emotive insight automatically removes them.

## B- Feature Extraction

In this work, Wavelet transform, the same method in was used for feature extraction. To capture the Alpha and Beta band waves, 4-level decomposition of the EEG signals must be executed. The common features captured were as follows:

• Energy of wavelet coefficient: the energy is extracted using the equation

$$E = \sum_{i=1}^{N} |D_i|^2 \dots (14)$$

Where D<sub>i</sub> is the detailed coefficient corresponding to Level N.

• Mean: the average of signals in the band.

$$\mu = \frac{1}{N} \sum_{i=1}^{N} (D_i - \mu)^2 - \dots - (15)$$

• Other features such as variance, skewness, kurtosis, max and min.

## C- Classification

This stage comes after the extractions of features of the EEG signals that accomplished in the previous step. For each segment of the EEG, the feature vector needs to be labeled for mental commands (i.e. Push=1, Pull=2, etc.) to train the classifier to detect them and use it to control the robotic arm. The main aim of this is to make the subject brain get used to the device and second to make the brain produce the same patterns of signals for a specific action. For example, the first training is for pushing.

motor imagery, when the subject is imagining pushing an object. After several trials the brain will produce the same pattern of signals for such action. The Emotive software comes with experiment features for subject training as shown in Figure 36. From the figure we can see the subject sitting infront of the screen where there is a cube in the middle.



Figure (36) Subject Training

The subject has to relax and focus on the cube to perform several actions, such as Push, Pull, Lift, Drop, move right and move left. All the signals from the different trainings will be stored and preprocessed for artifact removal, feature extractions and classifications. Figure 37 shows the features (Energy and mean) extracted from Beta and Alpha waves for four mental commands.



Figure (37) Mean and Energy for EEG signal

The Arduino code uses the stored signals as a reference to convert them to commands to control the robotic arm. The result of this work is divided into two parts:

- 1. Separate movement.
- 2. Full pick and place movement.

Both of the parts were executed in two conditions, the Quiet mode, where there is no extra noise in the environment. The second condition is where there is noise in the environment, as people talking or playing music next to the subject.

1- Separate movements (Mental Command)

Each joint in the robotic arm is connected to a servomotor. The servomotor moves in two directions. That means each movement needs 1 mental command. For example, mental command move right will result in rotating the servo 90 degrees CW. And the mental command move left will force the motor to rotate 90 degrees CCW as shown in Figure 38 below. The movement of the Base joint was chosen for this section which also is shown in Figure 39 a, b, c.



### Figure (38) Mental Command

As it shown in Figure 39, the initial position of the robot arm was in figure b at 0 degree the figure a is moving right CW and figure c is rotating Left CCW. Beta waves are related with active thinking, problem solving, concentration, and increased attentiveness



А

Figure (39) Robot Arm (Base) Movement

В

In an agitated environment, beta waves rise. This is a method of being attentive in the face of imagined threats from ambient noise. Figures 40 and 41 show the beta waves amplitude changing as Figure 40 represents the beta waves in a quiet environment and the low beta waves are increasing.



Figure (40) Low Beta Waves in A Quiet Environment

Figure 41 shows the High beta waves in a noisy environment a music was playing next to the subject. From Figure 41, high beta waves were arising as the subject was in a high focus as the noise environment affected the training. The same results were achieved for the Robotic arm movement.



Figure (41) High Beta Waves in A Noisy Environment.

The execution accuracy is calculated regarding each joint. The number of correctly executed actions to the total number of actions attempted as shown in equation 16. Table 6 shows the execution accuracy for each joint of the robotic arm using mental commands.

Execusion Accuracy =  $\frac{No.of \ correctly \ excuted}{Total \ No \ of \ action \ attempt} * 100-----(16)$ 

Table (6) Execution Accuracy for Robotic Arm using HITI brain Software

Joint	Command	(	Quiet Environme	nt	Noisy Environment			
		Total Number of trials	Number of Correctly Executed Trial	Accuracy	Total Number of trials	Number of Correctly Executed Trial	Accuracy	
Base	Push (C.W)	10	10	100	10	9	90	
	Pull (C.C.W)	10	10	100	10	8	80	

Shouldor	Lift (Up)	10	9	90	10	9	90
Siloutuei	Drop (Down)	10	10	100	10	9	90
Filesw	Lift (Up)	10	9	90	10	7	70
ELDOW	Drop (Down)	10	9	90	10	7	70
Wrist Vortical	Right (UP)	10	10	100	10	7	70
whist verticat	Left (Down)	10	10	100	10	7	70
Wrist	Left (C.W)	10	9	90	10	8	80
Rotational	Right (C.C.W)	10	10	100	10	8	80
Manipulator	Push (Open)	10	10	100	10	9	90
Manipulator	Pull (Close)	10	10	100	10	9	90

From the table above, some of the joints have the same mental commands that is due to the limitations of the Emotive software for the subscriptions this research could afford. And since the testing was for each joint separately, the testing can be performed with the same mental command. Figure 42 shows the difference of the execution accuracy in both conditions.



Figure (42) Execution Accuracy (HITI Brain)

2- Separate movement (Facial Expression).

The BCI records the signals of the brain based on special facial actions. Every time a person winks, blinks, smiles or does special action in the face, the brain produces different type of signals. Moreover, since the servomotor rotates in two directions, each direction requires one facial action. For example, a smile will force the motor to rotate CW. In addition, the raised eyebrows facial action leads to rotating the motor CCW. Figure

43 a, b shows the facial expression. In this section, the accuracy calculated was for the command recognition accuracy which can be calculated using Equation 17. Table 7 shows the command recognition accuracy for the facial recognition.



Figure (43) Facial Expression Command Recognition Accuracy =  $\frac{No.of \ correctly \ identified}{Total \ No \ of \ command \ sent} * 100$ -----(17)

Table (7)	Command	Recognition	Accuracy

	Qı	uiet Environment		Noisy Environment				
Command	Total Number of trials	Number of Correctly Executed Trial	Accuracy	Total Number of trials	Number of Correctly Executed Trial	Accuracy		
Smile	10	10	100	10	9	90		
Blink	10	10	100	10	10	100		
raise eyebrow	10	8	80	10	9	90		
wink left	10	10	100	10	8	80		
wink right	10	10	100	10	8	80		

clench teeth	10	10	100	10	9	90
furrow eyebrows	10	7	70	10	7	70

The accuracy of facial recognition (as shown in figure 44) in both conditions (Quiet and Noisy Environment) was high as this depends on reading the signals from the brain based on physical actions, there was no motor imagery in this test. Figure 45 a, b, c shows movement of the elbow joint that was chosen for this section.



Figure (44) Command Recognition Accuracy (Facial Expression)





From the Figures above, Figure 'b' represents the zero position, Figure 'a' is for CCW shoulder joint and Figure 'c' shows the CW rotation of the shoulder joint.

3- Full Pick and Place movement.

This is a little hard to perform since the work is limited by the actions available in the emotive software. The robotic arm has 6 joints and each joint moves in two directions. So, the total brain actions needed for the full arm are 12 actions. Therefore, in order to perform this complex movement with the limited resources in hands, the combinations of mental commands and facial expression were implemented in this section. Table 8 presents each joint with their movement and which action is connected to it. Table 9 and figure 46 present the execution accuracy for combined movement of different joints at the same time. It is crucial to mention that the test was performed in a quiet environmental condition only.

Joint		CW rotation	CCW rotation	Mental Command	Facial Expression
Base		Move Right	Move Left	Yes	No
Shoulder		Push	Pull Yes		No
Elbow		Lift	Drop	Yes	No
	Rotational	Raise brows	Furrow brows No		Yes
Wrist	Transitional	Wink left	Wink Right	No	Yes
Manipulator		Smile (Open)	Clench Teeth (Close)	No	Yes

### Table (8) Joints and Brain Actions.

#### Table (9) Execution Accuracy for Combined Joint Movement

	Quiet Environment							
Joint	Commands	Total Number of trials	Number of Correctly Executed Trial	Accuracy				
Base (CCW)-shoulder(Up)- Elbow(Down)	Left-Push-Drop	10	8	80				
Elbow(Up)-shoulder(Down)- Base(CW)	Lift-Pull-Right	10	8	80				





Figure (46) Execution Accuracy for Combined Joints Movement

Figure 47 (a-f) shows how full movement of an object is performed from one point to another









d





Figure (47) Pick and Place (Cont'd).

# 4- Safety and Security.

The system has two safety conditions in order to avoid colliding with other objects on the way and to avoid if the signal is too powerful that makes the joint move more than the desired motion. The first solution is to set the minimum and maximum interval range for the servo. This can be done in the code. The second solution is in the HITI brain software (Figure 48). When linking the joint movement to brain signals, there is an option called a threshold. This can determine the power the signal should reach before moving the motor. The safety of this system is very crucial to ensure smooth and safe working of the arm, as the system is also applicable for a prosthetic arm with 6 joints. And since the prosthetic attached to the body is different than the robotic arm, putting the safety conditions is required always.

HITIBrain (2_	_Braccio_Keyboa	ard)											- 0	· >
File Tools Lice	ense Help													
$+ \square$	B k		I II	635								HEAD	IO DATA	HOM
EMOTIV Co					Headset: 🕻	-					• : 📼 👘			
ARDUINO SK	ketch : Open		COM4	Open 🤇	Board :									
												4		
			Linker - Ard	uino						100	×			
				EMOTIV HE	ADSET					ARDUINO				
				-								LINK		
				Mental C	ommand					igital Data		LINK		
				Push				Address	a	sta 0		LINK		
												LINK		
												LINK		
				er 📃	0%							LINK		
						_•	📫 💼 Enable Un					LINK		
		7	VVrist Ve	r : Down	LINK									
		8	Wrist Ro		LINK									
			Wrist Ro		UNIS									
					< A									

Figure (48) HITI Software Safety Option

4.3. Controlling a higher degree of freedom robotic arm with few mental commands.

One of the limitations in the previous sections is that the more degree of freedom for the robot the more mental commands are required to control it. For example, as in the previous section, 6 DOF robotic arm requires 12 mental command or 12 different type of brain signals. Due to that reason, the patient will require more training. To solve this problem, we can use four mental commands, two mental commands to control the movement of each joint, and two mental commands to control the transition between the joints. For this solution, Node-RED software was used as well as the Arduino software and Emotive software.

## 4.3.1. Node-RED

Node-RED is one of the flow-based programming tools. The flow programming described the application's behavior as a black box network. Node-RED describes it as nodes [94]. Processing is defined in each node; data is given to it, processing is performed using that data, and that data is passed to the next node. The network plays the role of allowing data to flow between the nodes (Figure 49). Node-RED has the ability to combine the two-software necessary for this work (the Emotive and the Arduino). Node-RED basically is a programming tool for programming node.js applications with Graphical User Interface GUI tools [95].



Figure (49) Node-RED

The human arm has a wide range of motion, starting at the shoulder and extending to the elbow, wrist, and hand [96]. Joint and muscle group contributes to arm movement, beginning with the shoulder:

# 1. Shoulder Joint.

The shoulder is a ball-and-socket joint composed of the humerus (upper arm

bone) and the scapula (shoulder blade), especially the glenoid cavity [97].

This joint provides a wide range of motion [98], including:

- Flexion involves raising the arm forward.
- Extension: Moving the arm rearward.
- Abduction means lifting the arm horizontally away from the body.
- Adduction is bringing the arm back toward the body.
- rotation: internal (inward) and exterior (outward). Rotation

• Circumduction is the combination of the above actions that move the arm in a circular pattern.

Muscles Involved

- Deltoid: This muscle caps the shoulder and is responsible for abduction.
- The rotator cuff muscles (supraspinatus, infraspinatus, teres minor, and subscapularis) support the shoulder joint and help with numerous motions, notably rotation and abduction.
- The Pectoralis Major and Latissimus Dorsi muscles help with arm flexion, adduction, and rotation.
- Trapezius and Serratus Anterior: Stabilize the scapula and facilitate shoulder mobility.

# 2. Elbow joint.

The elbow is a hinge joint that links the humerus, ulna, and radius in the forearm [99].

It mainly allows:

- Flexion involves bending the arm.
- Extension: straightening the arm. Muscles Involved
- Biceps Brachii: The muscle responsible for elbow flexion.
- Triceps Brachii: The muscle responsible for elbow extension.

# 3. Forearm Rotation.

- This includes the proximal and distal radioulnar joints [99], which allow for:
- Pronation: Turning the palm downwards.
- Supination involves turning the palm upward.

Pronation is controlled by the Pronator Teres and Pronator Quadratus muscles. Supination requires the Biceps Brachii and Supinator muscles.

# 4. Wrist and Hand Joints.

The wrist and hand have multiple joints, enabling fine motor control and complex movements such as flexion, extension, abduction, adduction, and circumduction of the wrist, as well as grasping and manipulating objects with the fingers[100].

Muscles Involved:

• Flexor and Extensor Groups: These are found in the forearm and control movement at the wrist and fingers.
• Intrinsic Hand Muscles: Small muscles within the hand provide precise finger motions.

The coordination of these joints and muscles allows the human arm to perform a wide range of complex movements, from reaching and lifting to fine motor skills like writing. The central nervous system (CNS) plays a crucial role in controlling these movements by sending signals to the muscles, allowing for smooth, coordinated actions. From all above, the idea of controlling the robotic arm with less mental command. Each joint requires two movements, the clockwise and the counterclockwise movement. These two can be controlled with two mental commands. The transition between the joints can be controlled in two mental commands. One command is to transfer to the next joint and one is to back to the previous joint. So, to perform the full pick and place, the movement starts to move the shoulder joint to the required position, followed by the elbow, wrists and manipulator. The Node-Red flow blocks are shown in the figure 50. There are five mental commands (Push, Pull) for the CW and CCW movement, (Lift, Drop) for the transition between the joints. Neutral for stop command.





Table 10 presents the execution accuracy of switch control between different joints. To evaluate the system's robustness, tests were performed in both quiet and noisy environments, simulating varying real-world conditions. The quiet environment aimed to minimize external distractions and enhance focus, while the noisy environment introduced auditory and visual disturbances to test the system's reliability under challenging conditions. These experiments aimed to assess the feasibility and accuracy of controlling a robotic arm with limited mental commands in diverse scenarios.

Quiet	Environme	ent	Forward movement		Noisy Environment			Forward movement	
Joints movemen t	Command	No. of Trials	No. of True Excution	Accuracy	Joints movemen t	Comman d	No. of Trials	No. of True Excution	Accuracy
Base- Shoulder	Lift	10	10	100	Base- Shoulder	Lift	15	11	73.33333333
Base to Elbow	Lift	10	8	80	Base to Elbow	Lift	15	10	66.66666667
Base-Wrist Vertical	Lift	10	8	80	Base-Wrist Vertical	Lift	15	10	66.66666667
Base-Wrist Rotational	Lift	10	7	70	Base-Wrist Rotational	Lift	15	10	66.66666667
Base- Manipulator	Lift	10	8	80	Base- Manipulator	Lift	15	8	53.33333333
Quiet Environment			<b>.</b>		Noisy Environment			<b>Backward Movement</b>	
Quiet	Environme	ent	Backward	Movement	NoisyE	Invironm	ent	Backwa	rd Movement
Quiet Joints movemen t	Command	No. of Trials	Backward No. of True Excution	Movement Accuracy	Noisy E Joints movemen t	Comman d	ent No. of Trials	Backwa No. of True Excution	rd Movement Accuracy
Quiet Joints movemen t Shoulder- base	Command Drop	No. of Trials	Backward No. of True Excution	Accuracy 100	Noisy E Joints movemen t Shoulder- base	Comman d Drop	ent No. of Trials	Backwa No. of True Excution	Accuracy 66.66666667
Quiet Joints movemen t Shoulder- base Wrist Rotational- shoulder	Environme Command Drop Drop	No. of Trials	Backward No. of True Excution 10 7	Accuracy 100 70	NOISY E Joints movemen t Shoulder- base Wrist Rotational- shoulder	Drop	ent No. of Trials 15 15	Backwa No. of True Excution 10 8	rd Movement Accuracy 66.66666667 53.33333333
Quiet Joints movemen t Shoulder- base Wrist Rotational- shoulder Manipulator- base	Environme Command Drop Drop	No. of Trials 10 10 10	No. of True Excution 10 7 5	Accuracy 100 70 50	NOISY E Joints movemen t Shoulder- base Wrist Rotational- shoulder Manipulator- base	Drop Drop	ent No. of Trials 15 15	Backwa No. of True Excution 10 8 8	rd Movement Accuracy 66.666666667 53.33333333 53.33333333
Quiet Joints movemen t Shoulder- base Wrist Rotational- shoulder Manipulator- base Manipulator- Wrist Vertical	Environme Command Drop Drop Drop	No. of        Trials        10        10        10        10        10	Backward No. of True Excution 10 7 5 7	Movement        Accuracy        100        70        50        70	NOISY E Joints movemen t Shoulder- base Wrist Rotational- shoulder Manipulator- base Manipulator- Wrist Vertical	Drop Drop Drop	ent No. of Trials 15 15 15 15 15	Backwa No. of True Excution 10 8 8 8	rd Movement Accuracy 666.66666667 53.33333333 53.33333333 60

Table (10) Execution Accuracy of Joint Switch Control

In figures 51, 52 we can observe that there is a noticeable difference of execution accuracy between the Quiet environment and the Noisy environment.



Figure (51) Execution Accuracy of Forward Transition Between Joints



Figure (52) Execution Accuracy of Backward Transition between Joints

Table 11 and figure 53 show the combined switch control between joints at the same time. In this test, we only performed it in a Quiet environment condition.

## Table (11) Execution Accuracy of Combined Movement

Joints movement	No. of Trials	No. of True Excution	Accuracy
Base-Shoulder-Base	20	19	95

Shoulder-Elbow-Base	20	17	85
Wrist Vertical-Wrist-Rotational- Manipulator-Base	20	14	70



Figure (53) Execution Accuracy of Combined Movement

The more complicated the operation is the more focused and training the patient needs. It is important to mention that the execution of the robot arm in Pick and Place movement was applied in a quiet environment condition.

#### 4.4. Conclusion

Disability problem is widespread in the last decade. Paralyzed or people with lost limbs (especially the upper limb) face a real challenge performing daily life activities. The aim of this research is to help them to perform simple daily life activities using their brain signal. The work shows great results regarding the motion control of the robotics arm, and at the same time the system is safe and smooth. The patient before using the system needs to be trained for a period of time, and this time is different from person to person depends on power of focus in the patient. The overall system can be applicable for prosthetics. This finding supports **Hypothesis 3:** Employing a Brain-Computer Interface (BCI) for controlling a 6 Degrees of Freedom (DOF) robotic arm that closely replicates human arm movements facilitates more intuitive and precise robotic control. And **Hypothesis 4:** An innovative technique to control higher degrees of freedom of a robotic arm with reduced cognitive commands from the brain can improve usability and efficiency.

#### 5. BCI control for wheeled robot

Brain-controlled robots capture and use brain wave data to control movement. This, when combined with a mobile robot or wheelchair for handicapped people who cannot talk or move their hands, will allow them to move around independently [101]. To control the wheelchair, EEG signals are required. In this technique, the intention is to employ five electrodes to record EEG signals from various positions on the head. A Brain-Computer Interface (BCI) typically operates mobility robots or electric wheelchairs via Bluetooth. A BCI includes three components: input (e.g., user electrophysiological activity), output (i.e., device commands), and operation. Electrodes on the scalp or in the skull collect brain signals and extract certain properties, such as evoked potential amplitudes, sensory-motor cortex rhythms, and cortical neuron firing rates, to determine the user's intent. These characteristics are turned into commands that control a device (such as a mobile robot, wheelchair, or neuro-prosthesis [102]. The success of BCI operation is dependent on the interplay of two adaptive controllers: the user and the system. The user must create and maintain a strong link between his or her purpose and the signal characteristics used by the BCI, and the BCI must choose, and extract features that the user can influence before appropriately and effectively translating those features into device commands.

#### 5.1. Material

This section presents the resources utilized in the creation of BCI system. The components include the Emotive insight headset, the development software, the communication system, and the mobile robot. The Emotive Insight is a multi-sensory headset including 5 electrodes and two reference sensors. It acquires the EEG signals from the user's brain. Initially, Emotive Insight was introduced as an innovative control mechanism for video game interactivity. The mobile robot used in this work is Alphabot 2.0 (figure 54). It is an advanced robotic development platform designed by Waveshare. The robot is compatible with different types of controllers such as Arduino and Raspberry Pi.



Figure (54) Alphabot

#### 5.1.1. Mobile Robot.

A mobile robot is a software-controlled machine that uses sensors and various technologies to perceive its environment and navigate within it. Mobile robots operate by a synthesis of artificial intelligence (AI) and mechanical components, including wheels, tracks, and legs. Mobile robots are gaining prominence in several commercial areas [103]. They are employed to facilitate work processes and execute jobs that are infeasible or

hazardous for human laborers. It is possible to categorize mobile robots according to two criteria: the working environment and the means of propulsion. A few examples of mobile robots that can adapt to their surroundings are:

- Arctic robots that can navigate rough, icy terrain.
- Flying robots, often called drones or unmanned aerial vehicles (UAVs).
- Robots that move about on dry land or inside homes are known as unmanned ground vehicles (UGVs).
- Robots that can navigate themselves through water, also known as autonomous underwater vehicles (AUVs).
- Transportation and delivery robots are movable machines that can convey goods and supplies from one location to another.

A mobile robot may be classified using many devices, such as:

- legs, which can be either animal-like or human-like
- wheels.
- Tracks

In addition, autonomous mobile robots and non-autonomous mobile robots are the two most common varieties. When it comes to moving around, non-autonomous mobile robots need some type of instruction or guidance system, whereas non-autonomous mobile robots (AMRs) can navigate and discover their environment on their own [104]. A driving system called differential drive is used by several mobile robots. Each of its two wheels may be turned in the direction by itself; these wheels are positioned on a shared axis. Although the robot can roll by changing the velocities of its individual wheels, it can only do so by rotating around a point that lies along the axes of its two front wheels (Figure 55). The Instantaneous Center of Curvature (ICC) is the point around which the robot revolves [105]. By varying the velocity of each wheel, the trajectories taken by the robot can be changed accordingly.



Figure (55) ICC of Mobile Robot.

#### 5.1.2. Method of controlling the mobile robot.

A mobile robot known as a differential wheeled robot can move about thanks to its two wheels, one on each side of the body, which are controlled independently. It doesn't need any extra steering motion because it can change its direction just by adjusting the respective speeds of its wheels [106]. To avoid tipping over, robots using this type of propulsion usually incorporate one or more caster wheels. The robot will go in a straight path if the two wheels are turned at the same speed and in the same direction. The robot will spin around its axis if its two wheels are moved in opposing directions at the same speed. The point where the tires make contact is not always the center of rotation; rather, it can fall anywhere along that line, depending on the direction and speed of rotation. Even though the robot is moving in a straight path, the distance to the center of rotation is unlimited [107]. Accurate sensing and control of the speed and direction of rotation of the two driven wheels is crucial for the robot's direction. In this work, the Arduino controller is used to control Alphabot 2.0. With the combination of Node-Red, Arduino IDE and Emotive software, the control of Alphabot was achieved. Based on mental commands (Push, Pull, Right, Left, Neutral) the robot was moved (Forward, Backward, turn right, Turn Left, Stop) respectively. The Emotive Software sends the mental commands to Node-Red which acts as a medium that transfer the signals to the Arduino controller, and according to the code in the controller the movement of the mobile robot is executed. Figure (56) shows the flow diagram in Node-RED.



Figure (56) Node-RED Flow for Mobile Robot Control

#### 5.2. Results

The simplest way to make the mobile robot reach the goal point is to guide it. There are various methods to achieve that. In this work, the guidance using mental commands was used (Figure 57). The robot has to move between two points. The start point (A) and the end point (B). As in the previous chapter, the mobile robot was given instructions with brain signals to move in different directions. Figure 58 a, b, c, d shows how the mobile robot moves between points A and B. The subject sits in a relaxed state focusing on the robot movement. Five commands were used (Push= Forward, Pull = Backward, Left = Turn Left, Right = Turn Right, Neutral = Stop). It's important to mention that the subject had a training for 5 sessions and 40 trials, before testing.



Figure (57) Controlling Mobile Robot using BCI



Figure (58) Mental Commands Robot Controlling

The Execution accuracy was calculated as the same as the robotic arm. The ratio of the number of correctly executed actions to the total number of actions attempted. Table 12 and figure 59 present the results of the accuracy of the systems in the two conditions, the Quiet and noisy environment.

## Table (12) Execution Accuracy of Mobile Robot

Quiet Environment							
Command	No. of trials	No. of corrected executed	Accura cy	Command	No. of trials	No. of corrected executed	Accuracy
Forward - Right	10	10	100	Forward - Right	10	8	80
Forward - Left	10	10	100	Forward - Left	10	8	80
Backward-Right	10	9	90	Backward- Right	10	7	70
Backward-Left	10	10	100	Backward-Left	10	8	80
Full Move	10	8	80	Full Move	10	7	70





The results for the mobile robot were higher than in robotic arm as the robot doesn't have complex movements as the arm.

#### 5.3. Conclusion

The Emotive Insight BCI, Arduino, and Node-RED were successfully integrated to drive a mobile robot, demonstrating the possibility and usefulness of employing EEG-based brain signals for real-time robotic control. This technique demonstrates BCI technology's ability to provide intuitive and direct control of robotic devices in the absence of conventional interfaces. By mapping mental instructions to robot motions using Node-RED, I was able to obtain responsive and accurate control, demonstrating the utility of non-invasive EEG for mobile robotic applications. This experiment also lays the path for future advancements in BCI-controlled robots, with potential applications in assistive technology and beyond. This work supports **Hypothesis 5**: Using Brain-Computer Interface (BCI) technology with a fewer number of recording channels for the controlling of mobile robots will facilitate efficient and effective robotic control. Optimizing signal collection and processing with fewer channels enables accurate command execution while decreasing the complexity of the BCI system.

#### 6. CONCLUSION

The use of EEG signals from the brain to operate robotic systems, including robotic arms and mobile robots, represents a significant progression in assistive technology and brain-computer interface (BCI) applications. The statistics of Hungary, indicating that around 10% of the population is listed as handicapped, emphasize the social significance of this study. Assistive technology, such as robotic arms and mobile robots, operated using non-invasive EEG impulses, may provide significant answers to the issues encountered by this demographic. Prosthetic robotic arms operated by brainwaves might restore functionality and enhance autonomy for persons with motor disabilities, while brain-controlled mobile robots could facilitate movement and everyday activities, providing more freedom and dignity. The incorporation of EEG-controlled robotic systems signifies a crucial advancement in assistive technology for those with impairments. The results not only fulfill a significant social need, as shown by Hungary's disability statistics, but also facilitate worldwide progress in human-robot interaction. Through ongoing innovation in this domain, researchers and engineers may foster a more inclusive society, wherein technology enables people to surmount physical constraints and attain more autonomy in their lives.

## 7. New Scientific Results

The following theses encapsulate the essence and significance of my scientific research conducted during my Ph.D. studies:

## Thesis No. 1

I have demonstrated the practicality, safety, and applicability of non-invasive BCIs in real-world robotic applications, with a particular emphasis on assistive technologies for the disabled users. (Chapter 9/9.1 [3],[5], [??])

## Discussion

The following important elements are emphasized in my statement:

- 1. Real-World Applications: Non-invasive BCIs are more practicable and easier to use, making them more suitable for deployment in practical, everyday contexts. This is consistent with the requirements of incapacitated users, who need systems that are both user-friendly and dependable.
- 2. Safety Benefits: Non-invasive BCIs mitigate the health risks and complications that are linked to invasive and semi-invasive technologies, such as long-term maintenance, infection risks, and surgical procedures. This renders them a more acceptable and secure option for a broad user base, particularly for assistive purposes.
- 3. Emphasize Disabled Users: The objective is to establish systems that prioritize safety, comfort, and simplicity of adoption for individuals with disabilities. These criteria are more effectively met by non-invasive BCIs than by their invasive counterparts.
- 4. Adoption and Accessibility: Non-invasive BCIs are more likely to be widely adopted due to the fact that they do not necessitate specialized expertise for setup and maintenance or complex medical procedures. This renders them more suitable for mass-market assistive devices.

I reaffirm the thesis that non-invasive BCIs are not only technologically viable but also ethically and socially preferable options for assistive robotic systems

## Thesis No. 2

I have proven the effectiveness of combining advanced feature extraction techniques with robust classification methods in EEG data analysis, specifically for applications in BCI-controlled robotic systems (Chapter 9/ 9.1 [1], [3]).

### **Discussion:**

The following is illustrated by my statement:

- 1. Wavelet Transforms for Feature Extraction: I have demonstrated that wavelet transforms are highly effective in the extraction of meaningful and relevant features from EEG signals. This is of particular significance due to the fact that EEG data are frequently complex and chaotic, necessitating sophisticated preprocessing methods to enhance the signal-to-noise ratio and emphasize patterns during classification.
- 2. ANFIS for Classification: I demonstrated that the accuracy of EEG data analysis is considerably improved by employing the Adaptive Neuro-Fuzzy Inference System (ANFIS) as a classification method. This implies that ANFIS is an exceptional candidate for the classification of mental commands from EEG signals, as it is well-suited for the interpretation of nonlinear and equivocal data.
- 3. Supervised Machine Learning Applicability: The fact that supervised machine learning algorithms can similarly improve classification accuracy demonstrates that a variety of advanced methods, in addition to ANFIS, can be effectively employed for the classification of EEG signals. This broadens the scope of my findings, demonstrating that the results are not constrained to a single approach but rather to a general framework that integrates sophisticated classifiers and feature extraction.
- 4. Improved Accuracy: Your research demonstrates that the integration of sophisticated preprocessing (wavelet transformations) with potent classifiers (e.g., ANFIS or supervised machine learning algorithms like SVM and neural networks) leads to a substantial increase in classification accuracy. This enhancement is essential for real-world applications that require precision and reliability in the interpretation of EEG signals.

#### Thesis No. 3:

I have demonstrated the practicality and efficacy of employing Brain-Computer Interfaces (BCIs) to attain intuitive and precise control of intricate robotic systems, specifically a six-degrees-of-freedom (DOF) robotic arm (Chapter 9/9.1 [2], [6].

#### **Discussion:**

The following important elements are emphasized in the statement:

- 1. Human Arm Movement Replication: Your demonstration of the ability of BCIs to reconcile the divide between human intention and robotic execution was achieved through the use of a 6-DOF robotic arm that closely resembles human arm movements. This results in a more intuitive and natural control of the robotic limb for the user.
- 2. Intuitive Control: The capacity to operate a robotic arm through mental commands enables users to operate it without the need for extensive physical interfaces or manual inputs. The interaction is simplified, and cognitive and physical distress are reduced, which is particularly beneficial for users with disabilities.

- 3. Robotic Control Precision: The robotic arm's high degree of accuracy demonstrates that BCIs are capable of translating mental commands into precise, fine-grained movements. This is essential for applications that necessitate delicate or coordinated duties, such as industrial automation, prosthetics, or assistive devices.
- 4. Developing BCI Applications: Your research underscores the potential of BCIs to evolve from basic robotic duties to more complex systems with greater degrees of freedom. You contribute to the body of evidence that supports the scalability and versatility of BCIs in robotics by demonstrating successful control of a 6 DOF robotic arm.
- 5. Practical Significance: The emphasis on intuitive and precision control demonstrates the practical applicability of this technology in real-world scenarios, particularly in assistive technologies for individuals with motor impairments or in environments that necessitate seamless human-robot interaction.

#### Thesis No. 4:

By using a simple approach, I demonstrated a new method to control a higher degree of freedom robotic arm with only four commands. Instead of assigning two mental commands to every joint in the robotic arm which leads (which leads to the fact that the higher DOF the more double Number mental commands required) only four mental commands used. With this method (Chapter 9/ 9.1 [8]).

### **Discussions:**

I demonstrated that:

- 1- Complex tasks can be managed by fewer mental commands.
- 2- This method could be applicable for different fields, like assistive devices.
- 3- The reduction in cognitive loads highlights the potential for reducing mental fatigue.

### Thesis No. 5:

I endorsed a statement emphasizing the prospective advantages of using Brain-Computer Interface (BCI) technology for the control of mobile robots. The assertion indicates that BCI may improve the efficiency and efficacy of robotic control by enabling direct brain inputs that control robot motions and commands. This integration may enhance accessibility and engagement with robots, particularly for those with physical disabilities (Chapter 9/9.1 [8]).

## 8. Future Work and Outlook

Following the successful deployment of Brain-Computer Interfaces (BCIs) for the control of a 6 Degrees of Freedom (DOF) robotic arm, several avenues for future study may be delineated to improve the application, usefulness, and efficiency of this technology.

Below is the list of possible developments to work on in the future:

## 1. Increasing Degrees of Freedom and Complexity

Future research may concentrate on enhancing control capabilities for robotic systems with more degrees of freedom or supplementary functionality, such as robotic hands exhibiting finger-level dexterity. This would enhance the system's capability to emulate the whole spectrum of human arm and hand motions, hence making it more appropriate for intricate motor tasks and sophisticated assistive applications.

## 2. Multimodal Interfaces:

Integrating BCIs with other control modalities, such as vocal commands, ocular tracking, or tactile sensors, may provide hybrid control systems. These systems would provide redundant control alternatives, enhancing their robustness and user-friendliness in complicated or high-stress situations.

## 3. Minimizing Reliance on Training.

Although existing BCI systems often need extensive user training, future endeavors may aim to diminish this need by creating more sophisticated classifiers or integrating transfer learning methodologies. This would improve the technology's accessibility for a wider array of users.

## 4. A generalization Among Users:

Creating systems that function efficiently for a varied array of users, irrespective of individual disparities in brain signal patterns, is a significant focus for future study. This may include developing adaptive brain-computer interfaces that can learn and adapt to individual users in real-time.

# 9. Author's Publications

## 9.1. Publications related to the research

[1] I. A. Satam, "EEG signal ANFIS classification for motor imagery for different joints of the same limb," *Military Technical Courier*, vol. 72, no. 1, pp. 330–350, 2024, doi: 10.5937/vojtehg72-46601.

[2] I. A. Satam and I. A. Satam, "Review Studying of the Latest Development of Prosthetic Limbs Technologies," *Int J Sci Eng Res*, vol. 12, 2021, [Online]. Available: http://www.ijser.org

[3] I. A. Satam and R. Szabolcsi, "Supervised machine learning algorithms for brain signal classification," *Military Technical Courier/Vojnotehnicki glasnik*, vol. 72, no. 2, pp. 727–749, Apr. 2024, doi: 10.5937/vojtehg72-48620.

[4] I. A. Satam "Safety and Security Aspects of Implanted Brain-Computer Interface (BCI) for Human-Robot Interaction," *International Journal of Multidisciplinary Research and Publications (IJMRAP)*, vol. 6, no. 6, pp. 268–276, 2023, [Online]. Available: https://www.researchgate.net/publication/376488488

[5] I. Satam, "A comprehensive study of EEG-based control of artificial arms," *Vojnotehnicki glasnik*, vol. 71, no. 1, pp. 9–41, 2023, doi: 10.5937/vojtehg71-41366.

[6] I. A. Satam and R. Szabolcsi, Implantable BCI: Ethical Concerns (Under Review ).

[7] I. A. Satam and R. Szabolcsi Neurofeedback-Driven 6-DOF Robotic Arm: Integration of Brain-Computer Interface with Arduino for Advanced Control (Under Review ).

[8] I. A. Satam and R. Szabolcsi Advanced Brain-Controlled Actuation: Harnessing EEG Signals for Multi-Actuator Coordination in Robotics (Under Review)

# 9.2. Further publications

[1] R. W. Daoud, O. M. Ali, O. K. Ahmed, and I. A. Satam, "Arduino-based design and implementation of experimental rooms with a trombe wall for solar cells applications," *Bulletin of Electrical Engineering and Informatics*, vol. 12, no. 3, pp. 1248–1255, Jun. 2023, doi: 10.11591/eei.v12i3.4522.

[2] S. Abdulrahman and M. Zaenal, "Design and implementation of a smart sterilizing device to solve the doorknob contamination problem," *Vojnotehnicki glasnik*, vol. 70, no. 3, pp. 680–695, 2022, doi: 10.5937/vojtehg70-37492.

[3] A. H. Ahmed, M. N. A. Al-Hamadani, and I. A. Satam, "Prediction of COVID-19 disease severity using machine learning techniques," *Bulletin of Electrical Engineering and Informatics*, vol. 11, no. 2, pp. 1069–1074, Apr. 2022, doi: 10.11591/eei.v11i2.3272.

[4] I. Satam, "Fuzzy-based smart system for controlling road lights," *Vojnotehnicki glasnik*, vol. 70, no. 2, pp. 297–313, 2022, doi: 10.5937/vojtehg70-36670.

## REFERENCES

- [1] K.-A. Mardal, M. E. Rognes, T. B. Thompson, and L. M. Valnes, *Mathematical Modeling of the Human Brain From Magnetic Resonance Images to Finite Element Simulation*, vol. 10. Simula SpringerBriefs on Computing, 2023. doi: https://doi.org/10.1007/978-3-030-95136-8.
- [2] D. Delisle-Rodriguez *et al.*, "Multi-channel EEG-based BCI using regression and classification methods for attention training by serious game," *Biomed Signal Process Control*, vol. 85, Aug. 2023, doi: 10.1016/j.bspc.2023.104937.
- [3] S. Choo, H. Park, J. Y. Jung, K. Flores, and C. S. Nam, "Improving classification performance of motor imagery BCI through EEG data augmentation with conditional generative adversarial networks," *Neural Networks*, vol. 180, Dec. 2024, doi: 10.1016/j.neunet.2024.106665.
- [4] N. Kaongoen, J. Choi, and S. Jo, "A novel online BCI system using speech imagery and ear-EEG for home appliances control," *Comput Methods Programs Biomed*, vol. 224, Sep. 2022, doi: 10.1016/j.cmpb.2022.107022.
- [5] H. Li *et al.*, "Facilitating applications of SSVEP-BCI by effective Cross-Subject knowledge transfer," *Expert Syst Appl*, vol. 249, Sep. 2024, doi: 10.1016/j.eswa.2024.123492.
- [6] Z. Zhang, M. Li, R. Wei, W. Liao, F. Wang, and G. Xu, "Research on shared control of robots based on hybrid brain-computer interface," *J Neurosci Methods*, vol. 412, Dec. 2024, doi: 10.1016/j.jneumeth.2024.110280.
- [7] Y. An, J. Wong, and S. H. Ling, "Development of real-time brain-computer interface control system for robot," *Appl Soft Comput*, vol. 159, Jul. 2024, doi: 10.1016/j.asoc.2024.111648.
- [8] L. Cheng, D. Li, G. Yu, Z. Zhang, and S. Yu, "Robotic arm control system based on brain-muscle mixed signals," *Biomed Signal Process Control*, vol. 77, Aug. 2022, doi: 10.1016/j.bspc.2022.103754.
- [9] Z. Huang and M. Wang, "A review of electroencephalogram signal processing methods for braincontrolled robots," Jan. 01, 2021, *KeAi Communications Co.* doi: 10.1016/j.cogr.2021.07.001.
- [10] B. Chavarría, R. Zucca, A. Principe, A. Sanabria, and R. Rocamora, "Rapid intravenous loading of brivaracetam during invasive and non-invasive video-EEG monitoring," *Epilepsy Res*, vol. 192, May 2023, doi: 10.1016/j.eplepsyres.2023.107145.

- [11] J. Männlin *et al.*, "Safety profile of subdural and depth electrode implantations in invasive EEG exploration of drug-resistant focal epilepsy," *Seizure*, vol. 110, pp. 21–27, Aug. 2023, doi: 10.1016/j.seizure.2023.05.022.
- [12] S. H. Park *et al.*, "Epidural grid, a new methodology of invasive intracranial EEG monitoring: A technical note and experience of a single center," *Epilepsy Res*, vol. 182, May 2022, doi: 10.1016/j.eplepsyres.2022.106912.
- [13] M. Sonoda *et al.*, "Long-term satisfaction after extraoperative invasive EEG recording," *Epilepsy and Behavior*, vol. 124, Nov. 2021, doi: 10.1016/j.yebeh.2021.108363.
- [14] A. Novotná, E. Ehler, and M. Mareš, "12. Prognostic significance of interictal epileptiform discharges during semi-invasive eeg monitoring in patients with hippocampal sclerosis: Effects of partial drug withdrawal and sleep," *Clinical Neurophysiology*, vol. 125, no. 5, p. e28, May 2014, doi: 10.1016/j.clinph.2013.12.048.
- [15] T. Nakamura *et al.*, "Non-invasive electroencephalographical (EEG) recording system in awake monkeys," *Heliyon*, vol. 6, no. 5, May 2020, doi: 10.1016/j.heliyon.2020.e04043.
- [16] M. Lahijanian, M. J. Sedghizadeh, H. Aghajan, and Z. Vahabi, "Non-invasive auditory brain stimulation for gamma-band entrainment in dementia patients: An EEG dataset," *Data Brief*, vol. 41, Apr. 2022, doi: 10.1016/j.dib.2022.107839.
- [17] I. A. Satam and I. Abdulrahman Satam, "Safety and Security Aspects of Implanted Brain-Computer Interface (BCI) for Human-Robot Interaction," *International Journal of Multidisciplinary Research and Publications (IJMRAP*), vol. 6, no. 6, pp. 268–276, 2023, [Online]. Available: https://www.researchgate.net/publication/376488488
- [18] T. Fedele, H. J. Scheer, G. Waterstraat, B. Telenczuk, M. Burghoff, and G. Curio, "Towards noninvasive multi-unit spike recordings: Mapping 1kHz EEG signals over human somatosensory cortex," *Clinical Neurophysiology*, vol. 123, no. 12, pp. 2370–2376, Dec. 2012, doi: 10.1016/j.clinph.2012.04.028.
- [19] I. Satam, "A comprehensive study of EEG-based control of artificial arms," *Vojnotehnicki glasnik*, vol. 71, no. 1, pp. 9–41, 2023, doi: 10.5937/vojtehg71-41366.
- [20] Z. An, F. Wang, Y. Wen, F. Hu, and S. Han, "A real-time CNN-BiLSTM-based classifier for patientcentered AR-SSVEP active rehabilitation exoskeleton system ☆," 2024, doi: 10.13039/501100001809.

- [21] A. Akanmu, A. Okunola, H. Jebelli, A. Ammar, and A. Afolabi, "Cognitive load assessment of active back-support exoskeletons in construction: A case study on construction framing," *Advanced Engineering Informatics*, vol. 62, Oct. 2024, doi: 10.1016/j.aei.2024.102905.
- [22] L. Liu *et al.*, "Multimodal brain-controlled system for rehabilitation training: Combining asynchronous online brain-computer interface and exoskeleton," *J Neurosci Methods*, vol. 406, Jun. 2024, doi: 10.1016/j.jneumeth.2024.110132.
- [23] Y. N. Chen, Y. N. Wu, and B. S. Yang, "The neuromuscular control for lower limb exoskeleton- a 50-year perspective," Sep. 01, 2023, *Elsevier Ltd*. doi: 10.1016/j.jbiomech.2023.111738.
- [24] A. C. Villa-Parra, D. Delisle-Rodríguez, A. López-Delis, T. Bastos-Filho, R. Sagaró, and A. Frizera-Neto, "Towards a Robotic Knee Exoskeleton Control Based on Human Motion Intention through EEG and sEMGsignals," in *Procedia Manufacturing*, Elsevier B.V., 2015, pp. 1379–1386. doi: 10.1016/j.promfg.2015.07.296.
- [25] A. Rakshit, S. Pramanick, A. Bagchi, and S. Bhattacharyya, "Autonomous grasping of 3-D objects by a vision-actuated robot arm using Brain–Computer Interface," *Biomed Signal Process Control*, vol. 84, Jul. 2023, doi: 10.1016/j.bspc.2023.104765.
- [26] J. Seo *et al.*, "A phased array ultrasound system with a robotic arm for neuromodulation," *Med Eng Phys*, vol. 118, Aug. 2023, doi: 10.1016/j.medengphy.2023.104023.
- [27] M. Pan, M. O. Wong, C. C. Lam, and W. Pan, "Integrating extended reality and robotics in construction: A critical review," *Advanced Engineering Informatics*, vol. 62, Oct. 2024, doi: 10.1016/j.aei.2024.102795.
- [28] A. Ak, V. Topuz, and I. Midi, "Motor imagery EEG signal classification using image processing technique over GoogLeNet deep learning algorithm for controlling the robot manipulator," *Biomed Signal Process Control*, vol. 72, Feb. 2022, doi: 10.1016/j.bspc.2021.103295.
- [29] J. R. Williamson, D. W. Bliss, D. W. Browne, and J. T. Narayanan, "Seizure prediction using EEG spatiotemporal correlation structure," *Epilepsy and Behavior*, vol. 25, no. 2, pp. 230–238, Oct. 2012, doi: 10.1016/j.yebeh.2012.07.007.
- [30] A. Paul, "Epilepsy or stereotypy? Diagnostic issues in learning disabilities," *Seizure*, vol. 6, no. 2, pp. 111–120, 1997, doi: https://doi.org/10.1016/S1059-1311(97)80064-7.
- [31] E. Yıldırım, T. Aktürk, L. Hanoğlu, G. Yener, C. Babiloni, and B. Güntekin, "Lower oddball eventrelated EEG delta and theta responses in patients with dementia due to Parkinson's and Lewy

body than Alzheimer's disease," *Neurobiol Aging*, vol. 137, pp. 78–93, May 2024, doi: 10.1016/j.neurobiolaging.2024.02.004.

- [32] P. D. Barua, T. Tuncer, M. Baygin, S. Dogan, and U. R. Acharya, "N-BodyPat: Investigation on the dementia and Alzheimer's disorder detection using EEG signals," *Knowl Based Syst*, vol. 304, Nov. 2024, doi: 10.1016/j.knosys.2024.112510.
- [33] A. Ilyas and F. A. Rathore, "Comments on 'Gunshot injury to spine: An institutional experience of management and complications from a developing country'-----The need for an interdisciplinary spinal cord injury rehabilitation for improving outcomes in patients with gunshot injury to spine," *Chinese Journal of Traumatology - English Edition*, vol. 23, no. 6, pp. 329–330, Dec. 2020, doi: 10.1016/j.cjtee.2020.11.003.
- [34] S. M. Sarhan, M. Z. Al-Faiz, and A. M. Takhakh, "A review on EMG/EEG based control scheme of upper limb rehabilitation robots for stroke patients," Aug. 01, 2023, *Elsevier Ltd*. doi: 10.1016/j.heliyon.2023.e18308.
- [35] Y. Liu *et al.*, "A tensor-based scheme for stroke patients' motor imagery EEG analysis in BCI-FES rehabilitation training," *J Neurosci Methods*, vol. 222, pp. 238–249, Jan. 2014, doi: 10.1016/j.jneumeth.2013.11.009.
- [36] C. Chen, J. Lv, and Z. Xu, "A Multi-Indicator evaluation method for Human-Machine effectiveness of lower limb wearable exoskeleton," *Biomed Signal Process Control*, vol. 91, May 2024, doi: 10.1016/j.bspc.2024.105976.
- [37] P. Haselager, R. Vlek, J. Hill, and F. Nijboer, "A note on ethical aspects of BCI," *Neural Networks*, vol. 22, no. 9, pp. 1352–1357, Nov. 2009, doi: 10.1016/j.neunet.2009.06.046.
- [38] X. Li *et al.*, "Evaluation of an online SSVEP-BCI with fast system setup," *Journal of Neurorestoratology*, vol. 12, no. 2, Jun. 2024, doi: 10.1016/j.jnrt.2024.100122.
- [39] S. Rimbert and S. Fleck, "Long-term kinesthetic motor imagery practice with a BCI: Impacts on user experience, motor cortex oscillations and BCI performances," *Comput Human Behav*, vol. 146, Sep. 2023, doi: 10.1016/j.chb.2023.107789.
- [40] X. Li, W. Wei, S. Qiu, and H. He, "A temporal-spectral fusion transformer with subject-specific adapter for enhancing RSVP-BCI decoding," *Neural Networks*, vol. 181, p. 106844, Jan. 2025, doi: 10.1016/j.neunet.2024.106844.

- [41] A. Jain, R. Raja, S. Srivastava, P. C. Sharma, J. Gangrade, and M. R, "Analysis of EEG signals and data acquisition methods: a review," *Comput Methods Biomech Biomed Eng Imaging Vis*, vol. 12, no. 1, 2024, doi: 10.1080/21681163.2024.2304574.
- [42] U. Salahuddin and P. X. Gao, "Signal Generation, Acquisition, and Processing in Brain Machine Interfaces: A Unified Review," *Front Neurosci*, vol. 15, Sep. 2021, doi: 10.3389/fnins.2021.728178.
- [43] A. Bizzego and G. Esposito, "Acquisition and processing of brain signals," Sensors, vol. 21, no. 19, Oct. 2021, doi: 10.3390/s21196492.
- [44] G. Jeon, X. Yang, Y. Tian, and Y. Pang, "Editorial: Neural signals acquisition and intelligent analysis," *Front Neurosci*, vol. 17, 2023, doi: 10.3389/fnins.2023.1251280.
- [45] X. Lian *et al.*, "Causal influences of testosterone on brain structure change rate: A sex-stratified Mendelian randomization study," *Journal of Steroid Biochemistry and Molecular Biology*, vol. 245, Jan. 2025, doi: 10.1016/j.jsbmb.2024.106629.
- [46] S. Schöneich, "Neuroethology of acoustic communication in field crickets from signal generation to song recognition in an insect brain," *Prog Neurobiol*, vol. 194, Nov. 2020, doi: 10.1016/j.pneurobio.2020.101882.
- [47] T. F. Zaidi and O. Farooq, "EEG sub-bands based sleep stages classification using Fourier Synchrosqueezed transform features," *Expert Syst Appl*, vol. 212, Feb. 2023, doi: 10.1016/j.eswa.2022.118752.
- [48] S. Qammar and A. Khalid Khan, "BCI Systems and Comparison of Various Signal Acquisition Techniques," *Journal Data Sci Mod Tech 2022* |, vol. 6, no. 105, p. 105, 2022.
- [49] G. Torres, M. P. Cinelli, A. T. Hynes, I. S. Kaplan, and J. R. Leheste, "Electroencephalogram mapping of brain states," *J Neurosci Neuroeng*, vol. 3, no. 2, pp. 73–77, Dec. 2014, doi: 10.1166/jnsne.2014.1098.
- [50] J. P. Singh, "A set of five generalised memristive synapses for the hidden nonlinear dynamics in three coupled neurons," *Chaos Solitons Fractals*, vol. 188, Nov. 2024, doi: 10.1016/j.chaos.2024.115551.
- [51] A. Fiasconaro and M. Migliore, "Effects of synapse location, delay and background stochastic activity on synchronising hippocampal CA1 neurons," *Chaos, Solitons and Fractals: X*, vol. 13, Dec. 2024, doi: 10.1016/j.csfx.2024.100122.

- [52] K. Sekine, W. Haga, S. Kim, M. Imayasu, T. Yoshida, and H. Tsutsui, "Neuron-microelectrode junction induced by an engineered synapse organizer," *Biochem Biophys Res Commun*, vol. 712– 713, Jun. 2024, doi: 10.1016/j.bbrc.2024.149935.
- [53] B. J. Thio and W. M. Grill, "Relative contributions of different neural sources to the EEG," *Neuroimage*, vol. 275, Jul. 2023, doi: 10.1016/j.neuroimage.2023.120179.
- [54] H. Su *et al.*, "Analysis of brain network differences in the active, motor imagery, and passive stoke rehabilitation paradigms based on the task-state EEG," *Brain Res*, vol. 1846, Jan. 2025, doi: 10.1016/j.brainres.2024.149261.
- [55] X. Ma, S. Qiu, and H. He, "Multi-channel EEG recording during motor imagery of different joints from the same limb," *Sci Data*, vol. 7, no. 1, Dec. 2020, doi: 10.1038/s41597-020-0535-2.
- [56] M. H. Puglia, J. S. Slobin, and C. L. Williams, "The automated preprocessing pipe-line for the estimation of scale-wise entropy from EEG data (APPLESEED): Development and validation for use in pediatric populations," *Dev Cogn Neurosci*, vol. 58, Dec. 2022, doi: 10.1016/j.dcn.2022.101163.
- [57] S. Abenna, M. Nahid, H. Bouyghf, and B. Ouacha, "EEG-based BCI: A novel improvement for EEG signals classification based on real-time preprocessing," *Comput Biol Med*, vol. 148, Sep. 2022, doi: 10.1016/j.compbiomed.2022.105931.
- [58] S. Liu *et al.*, "MAS-DGAT-Net: A dynamic graph attention network with multibranch feature extraction and staged fusion for EEG emotion recognition," *Knowl Based Syst*, vol. 305, Dec. 2024, doi: 10.1016/j.knosys.2024.112599.
- [59] A. K. Singh and S. Krishnan, "Trends in EEG signal feature extraction applications," *Front Artif Intell*, vol. 5, 2022, doi: https://doi.org/10.3389/frai.2022.1072801.
- [60] 2012 9th International Conference on Fuzzy Systems and Knowledge Discovery, "Time domain Feature extraction and classification of EEG data for Brain Computer Interface," IEEE, 2012. doi: https://doi.org/10.1109/FSKD.2012.6234336.
- [61] V. K. B. Varsha K. Harpale, "Time and frequency domain analysis of EEG signals for seizure detection: A review," in 2016 International Conference on Microelectronics, Computing and Communications (MicroCom), IEEE, 2016. doi: https://doi.org/10.1109/MicroCom.2016.7522581.

- [62] R. Han, Z. Li, Y. Zhang, X. Meng, Z. Wang, and H. Dong, "Weighted common spatial pattern based adaptation regularization for multi-source EEG time series," *Computers and Electrical Engineering*, vol. 120, Dec. 2024, doi: 10.1016/j.compeleceng.2024.109680.
- [63] S. Nießner, M. Liewald, and M. Weyrich, "A methodology to quantify tool wear effects in a shear cutting process based on an automatic feature extraction," in *IFAC-PapersOnLine*, Elsevier B.V., 2022, pp. 540–547. doi: 10.1016/j.ifacol.2022.04.250.
- [64] H. Massar, T. Belhoussine Drissi, B. Nsiri, and M. Miyara, "Advancements in Blind Source Separation for EEG Artifact Removal: A comparative analysis of Variational Mode Decomposition and Discrete Wavelet Transform approaches," *Applied Acoustics*, vol. 228, Jan. 2025, doi: 10.1016/j.apacoust.2024.110300.
- [65] S. Bagherzadeh *et al.*, "A subject-independent portable emotion recognition system using synchrosqueezing wavelet transform maps of EEG signals and ResNet-18," *Biomed Signal Process Control*, vol. 90, Apr. 2024, doi: 10.1016/j.bspc.2023.105875.
- [66] P. D. Barua *et al.*, "Automated EEG sentence classification using novel dynamic-sized binary pattern and multilevel discrete wavelet transform techniques with TSEEG database," *Biomed Signal Process Control*, vol. 79, Jan. 2023, doi: 10.1016/j.bspc.2022.104055.
- [67] H. Ocak, "Automatic detection of epileptic seizures in EEG using discrete wavelet transform and approximate entropy," *Expert Syst Appl*, vol. 36, no. 2 PART 1, pp. 2027–2036, 2009, doi: 10.1016/j.eswa.2007.12.065.
- [68] A. Langenbucher, N. Szentmáry, J. Wendelstein, A. Cayless, P. Hoffmann, and D. Gatinel, "Performance Evaluation of a Simple Strategy to Optimize Formula Constants for Zero Mean or Minimal Standard Deviation or Root-Mean-Squared Prediction Error in Intraocular Lens Power Calculation," *Am J Ophthalmol*, vol. 269, pp. 282–292, Jan. 2025, doi: 10.1016/j.ajo.2024.08.043.
- [69] R. D. K. A. and S. I. Majid Aljalal, "Feature Extraction of EEG based Motor Imagery Using CSP based on Logarithmic Band Power, Entropy and Energy," in 1st International Conference on Computer Applications & Information Security : ICCAIS'2018 : Riyadh, Kingdom of Saudi Arabia, 04-06 April, 2018, IEEE, 2018. doi: https://doi.org/10.1109/CAIS.2018.8441995.
- [70] R. Yuvaraj, P. Thagavel, J. Thomas, J. Fogarty, and F. Ali, "Comprehensive Analysis of Feature Extraction Methods for Emotion Recognition from Multichannel EEG Recordings," *Sensors*, vol. 23, no. 2, Jan. 2023, doi: 10.3390/s23020915.

- [71] A. G. Mahapatra and K. Horio, "Classification of ictal and interictal EEG using RMS frequency, dominant frequency, root mean instantaneous frequency square and their parameters ratio," *Biomed Signal Process Control*, vol. 44, pp. 168–180, Jul. 2018, doi: 10.1016/j.bspc.2018.04.007.
- [72] K. Sheela Sobana Rani, S. Pravinth Raja, M. Sinthuja, B. Vidhya Banu, R. Sapna, and K. Dekeba,
  "Classification of EEG Signals Using Neural Network for Predicting Consumer Choices," *Comput Intell Neurosci*, vol. 2022, 2022, doi: 10.1155/2022/5872401.
- [73] H. Wang, B. Chen, H. Sun, A. Li, and C. Zhou, "ANFIS-MOH: Systematic exploration of hybrid ANFIS frameworks via metaheuristic optimization hybridization with evolutionary and swarm-based algorithms," *Appl Soft Comput*, vol. 167, Dec. 2024, doi: 10.1016/j.asoc.2024.112334.
- [74] A. H. Khalaf, B. Lin, A. N. Abdalla, Z. Han, Y. Xiao, and J. Tang, "Enhanced prediction of corrosion rates of pipeline steels using simulated annealing-optimized ANFIS models," *Results in Engineering*, vol. 24, Dec. 2024, doi: 10.1016/j.rineng.2024.102853.
- [75] D. Kim, J. Y. Park, Y. W. Song, E. Kim, S. Kim, and E. Y. Joo, "Machine-learning-based classification of obstructive sleep apnea using 19-channel sleep EEG data," *Sleep Med*, vol. 124, pp. 323–330, Dec. 2024, doi: 10.1016/j.sleep.2024.09.041.
- [76] M. Aljalal, M. Molinas, S. A. Aldosari, K. AlSharabi, A. M. Abdurraqeeb, and F. A. Alturki, "Mild cognitive impairment detection with optimally selected EEG channels based on variational mode decomposition and supervised machine learning," *Biomed Signal Process Control*, vol. 87, Jan. 2024, doi: 10.1016/j.bspc.2023.105462.
- [77] C. Xu, Y. Song, Q. Zheng, Q. Wang, and P. A. Heng, "Unsupervised multi-source domain adaptation via contrastive learning for EEG classification," *Expert Syst Appl*, vol. 261, Feb. 2025, doi: 10.1016/j.eswa.2024.125452.
- [78] M. Yousaf, M. Farhan, Y. Saeed, M. J. Iqbal, F. Ullah, and G. Srivastava, "Enhancing driver attention and road safety through EEG-informed deep reinforcement learning and soft computing," *Appl Soft Comput*, vol. 167, Dec. 2024, doi: 10.1016/j.asoc.2024.112320.
- [79] X. Shan and E. H. Yang, "Supervised machine learning of thermal comfort under different indoor temperatures using EEG measurements," *Energy Build*, vol. 225, Oct. 2020, doi: 10.1016/j.enbuild.2020.110305.
- [80] D. He, D. Ren, Z. Guo, and B. Jiang, "Insomnia disorder diagnosed by resting-state fMRI-based SVM classifier," *Sleep Med*, vol. 95, pp. 126–129, Jul. 2022, doi: 10.1016/j.sleep.2022.04.024.

- [81] Q. Wang *et al.*, "A hybrid SVM and kernel function-based sparse representation classification for automated epilepsy detection in EEG signals," *Neurocomputing*, vol. 562, Dec. 2023, doi: 10.1016/j.neucom.2023.126874.
- [82] X. Liao, G. Li, Y. Wang, L. Sun, and H. Zhang, "Constructing lightweight and efficient spiking neural networks for EEG-based motor imagery classification," *Biomed Signal Process Control*, vol. 100, Feb. 2025, doi: 10.1016/j.bspc.2024.107000.
- [83] X. Ma, S. Qiu, and H. He, "Multi-channel EEG recording during motor imagery of different joints from the same limb," *Sci Data*, vol. 7, no. 1, Dec. 2020, doi: 10.1038/s41597-020-0535-2.
- [84] M. van den Boom *et al.*, "Typical somatomotor physiology of the hand is preserved in a patient with an amputated arm: An ECoG case study," *Neuroimage Clin*, vol. 31, Jan. 2021, doi: 10.1016/j.nicl.2021.102728.
- [85] M. Tariq, P. M. Trivailo, and M. Simic, "Classification of left and right knee extension motor imagery using common spatial pattern for BCI applications," in *Procedia Computer Science*, Elsevier B.V., 2019, pp. 2598–2606. doi: 10.1016/j.procs.2019.09.256.
- [86] S. López Bernal, J. A. Martínez López, E. T. Martínez Beltrán, M. Quiles Pérez, G. Martínez Pérez, and A. Huertas Celdrán, "NeuronLab: BCI framework for the study of biosignals," *Neurocomputing*, vol. 598, Sep. 2024, doi: 10.1016/j.neucom.2024.128027.
- [87] A. K., N. S., Y. G., W. J. Priyanka Datta, "An evaluation of intelligent and immersive digital applications in eliciting cognitive states in humans through the utilization of Emotiv Insight," *MethodsX*, vol. 12, no. 11, Jun. 2022, doi: 10.3390/app12115499.
- [88] Z. Ali *et al.*, "Design and development of a low-cost 5-DOF robotic arm for lightweight material handling and sorting applications: A case study for small manufacturing industries of Pakistan," *Results in Engineering*, vol. 19, Sep. 2023, doi: 10.1016/j.rineng.2023.101315.
- [89] T. T. Tung, N. Van Tinh, D. T. Phuong Thao, and T. V. Minh, "Development of a prototype 6 degree of freedom robot arm," *Results in Engineering*, vol. 18, Jun. 2023, doi: 10.1016/j.rineng.2023.101049.
- [90] C. Relaño, J. Muñoz, and C. A. Monje, "Gaussian process regression for forward and inverse kinematics of a soft robotic arm," *Eng Appl Artif Intell*, vol. 126, Nov. 2023, doi: 10.1016/j.engappai.2023.107174.

- [91] D. Debnath, P. Malla, and S. Roy, "Position control of a DC servo motor using various controllers: A comparative study," in *Materials Today: Proceedings*, Elsevier Ltd, Jan. 2022, pp. 484–488. doi: 10.1016/j.matpr.2022.03.008.
- [92] B. Graimann, J. E. Huggins, S. P. Levine, and G. Pfurtscheller, "Visualization of significant ERD/ERS patterns in multichannel EEG and ECoG data," *Clinical Neurophysiology*, vol. 113, pp. 43–47, 2002, doi: https://doi.org/10.1016/S1388-2457(01)00697-6.
- [93] C. M. Krause, P. A. Boman, L. Sillanmäki, T. Varho, and I. E. Holopainen, "Brain oscillatory EEG event-related desynchronization (ERD) and -sychronization (ERS) responses during an auditory memory task are altered in children with epilepsy," *Seizure*, vol. 17, no. 1, pp. 1–10, Jan. 2008, doi: 10.1016/j.seizure.2007.05.015.
- [94] C. Y. Chen, S. H. Wu, B. W. Huang, C. H. Huang, and C. F. Yang, "Web-based Internet of Things on environmental and lighting control and monitoring system using node-RED, MQTT and Modbus communications within embedded Linux platform," *Internet of Things (Netherlands)*, vol. 27, Oct. 2024, doi: 10.1016/j.iot.2024.101305.
- [95] A. Sinan Cabuk, "Experimental IoT study on fault detection and preventive apparatus using Node-RED ship's main engine cooling water pump motor," *Eng Fail Anal*, vol. 138, Aug. 2022, doi: 10.1016/j.engfailanal.2022.106310.
- [96] G. Ramella, L. Grazi, F. Giovacchini, E. Trigili, N. Vitiello, and S. Crea, "Evaluation of antigravitational support levels provided by a passive upper-limb occupational exoskeleton in repetitive arm movements," *Appl Ergon*, vol. 117, May 2024, doi: 10.1016/j.apergo.2024.104226.
- [97] D. Kawaguchi, H. Tomita, Y. Fukaya, and A. Kanai, "A coactivation strategy in anticipatory postural adjustments during voluntary unilateral arm movement while standing in individuals with bilateral spastic cerebral palsy," *Hum Mov Sci*, vol. 96, Aug. 2024, doi: 10.1016/j.humov.2024.103255.
- [98] R. Dey, J. Glenday, J. P. du Plessis, S. Sivarasu, and S. Roche, "Characterizing moment arms of the coracobrachialis and short head of biceps in native shoulders and after reverse total shoulder arthroplasty during elevation and rotation: a modeling study," *J Shoulder Elbow Surg*, vol. 32, no. 7, pp. 1380–1391, Jul. 2023, doi: 10.1016/j.jse.2023.01.015.
- [99] R. Hosoda and G. Venture, "Human elbow joint torque estimation during dynamic movements with moment arm compensation method," in *Proceedings of the 19th World Congress The*

*International Federation of Automatic Control Cape Town, South Africa*, Aug. 2014. doi: https://doi.org/10.3182/20140824-6-ZA-1003.01266.

- [100] H. Fu, Y. Yu, Y. Chen, Z. Wang, and T. Wu, "Direct measurement for hand-transmitted vibration of human fingers-hand-arm system in the zero-gravity environment," *Advances in Space Research*, vol. 74, no. 8, pp. 3916–3924, Oct. 2024, doi: 10.1016/j.asr.2024.06.062.
- [101] T. I. Voznenko, E. V. Chepin, and G. A. Urvanov, "The control system based on extended BCI for a robotic wheelchair," in *Procedia Computer Science*, Elsevier B.V., 2018, pp. 522–527. doi: 10.1016/j.procs.2018.01.079.
- [102] J. Zhang and M. Wang, "A survey on robots controlled by motor imagery brain-computer interfaces," Jan. 01, 2021, *KeAi Communications Co.* doi: 10.1016/j.cogr.2021.02.001.
- [103] A. Top and M. Gökbulut, "Real-time deep learning-based position control of a mobile robot," *Eng Appl Artif Intell*, vol. 138, Dec. 2024, doi: 10.1016/j.engappai.2024.109373.
- [104] K. Ozdemir and A. Tuncer, "Navigation of autonomous mobile robots in dynamic unknown environments based on dueling double deep q networks," *Eng Appl Artif Intell*, vol. 139, Jan. 2025, doi: 10.1016/j.engappai.2024.109498.
- [105] L. Ye, F. Wu, X. Zou, and J. Li, "Path planning for mobile robots in unstructured orchard environments: An improved kinematically constrained bi-directional RRT approach," *Comput Electron Agric*, vol. 215, Dec. 2023, doi: 10.1016/j.compag.2023.108453.
- [106] L. Zhang, Z. Cai, Y. Yan, C. Yang, and Y. Hu, "Multi-agent policy learning-based path planning for autonomous mobile robots," *Eng Appl Artif Intell*, vol. 129, Mar. 2024, doi: 10.1016/j.engappai.2023.107631.
- [107] B. Moudoud and H. Aissaoui, "Fixed-time adaptive sliding mode-based trajectory tracking control for Wheeled Mobile Robot: Theoretical development and real-time implementation," *e-Prime - Advances in Electrical Engineering, Electronics and Energy*, vol. 10, p. 100830, Dec. 2024, doi: 10.1016/j.prime.2024.100830.